

Lending when relationships are scarce: The role of information spread via bank networks

Yan Alperovych, Anantha Divakaruni, Sophie Manigart

▶ To cite this version:

Yan Alperovych, Anantha Divakaruni, Sophie Manigart. Lending when relationships are scarce: The role of information spread via bank networks. Journal of Corporate Finance, 2022, 73, 10.1016/j.jcorpfin.2022.102181. hal-04325549

HAL Id: hal-04325549 https://hal.science/hal-04325549

Submitted on 22 Jul 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial 4.0 International License

Lending when Relationships are Scarce: The Role of Information Spread via Bank Networks*

Yan Alperovych[†], Anantha Divakaruni[‡], and Sophie Manigart[⊥]

December 13, 2022

Abstract

We investigate how information flows within bank networks facilitate syndicate formation and lending in the leveraged buyout (LBO) market, where relationships between banks and borrowers are scarce and borrower opacity is high. Using novel measures that characterize a bank's ability to source and disseminate information within its loan syndication network, we show that the extent of this capability influences which banks join the syndicate, the share the lead bank holds, and LBO borrowing terms. Banks' ability to source and disseminate network-based information is particularly useful when ties to prospective borrowers are lacking, with the information flows extending beyond knowledge on PE firms and LBO targets.

Keywords: Networks; Syndication; Lending; Information; Private Equity; Leveraged Buyouts

^{*} We thank Riccardo Calcagno, Hans Degryse, Paolo Fulghieri, Thomas Hellmann, Tim Jenkinson, Miguel Meuleman, Alan Morrison, and Mike Wright for helpful comments. We also thank participants of the 2017 Annual Meeting of the Financial Management Association, 2nd ENTFIN Conference, AFFI 2018 spring conference, and seminar participants at Emlyon business school, Ghent University, and Université de Lille.

[†] Email: alperovych@em-lyon.com; Address: Emlyon Business School, 23, Avenue Guy de Collongue, Ecully, 69134 France.

[‡] Email: anantha.divakaruni@uib.no; Address: University of Bergen, PO Box 7802, 5020 Bergen, Norway.

[⊥] Email: sophie.manigart@vlerick.com; Address: Vlerick Business School and Ghent University, Reep 1, Gent 9000 Belgium.

JEL	classifications:	D82,	G20,	G21,	G23,	G24,	L14
-----	------------------	------	------	------	------	------	-----

1. Introduction

A large body of research has investigated the benefits to banks of building close ties with borrowers. Using the duration and intensity of borrowing as proxies for lending relationships, this literature documents that strong ties reduce information asymmetries between banks and borrowers (Boot, 2000; Diamond, 1984; Farinha and Santos, 2002; Rajan, 1992; Sharpe, 1990), thereby improving credit access (Petersen and Rajan, 1994; 1995), reducing the cost of debt (Berger and Udell, 1995; Bolton, Freixas, Gambacorta and Mistrulli, 2016; Ivashina and Kovner, 2011; López-Espinosa, Mayordomo and Moreno, 2017), and lowering collateral requirements (Degryse and Van Cayseele, 2000).

While these studies showcase the benefits of relationship lending to both banks and borrowers, lending relationships are often scarce and difficult to establish in settings characterized by acute information asymmetries. One such important setting is the market for leveraged buyouts (LBOs) where a private equity (PE) firm uses mainly debt to acquire a "target" company.¹ The majority of LBO targets are private firms whose creditworthiness is hard to evaluate compared to publicly listed peers. Given that the median LBO in our data carries a debt load exceeding \$200 million and 5 years in loan maturity, it follows that the banks funding these deals are exposed to substantial credit risks (about \$50 million per bank in a syndicate on average).

Theory suggests that banks adopt two main strategies to deal with this important risk exposure. First, two or more banks might form a "syndicate" to share risks and jointly issue the LBO loan (Sufi, 2007). Second, banks will lend mostly to borrowers with whom they share strong ties to reduce risk and uncertainty (Ivashina and Kovner, 2011). In practice, while the syndication of LBO loans is widespread, our data shows that PE firms and targets do not have ties to banks in the LBO loan syndicate in at least 48% of the deals in our sample.² Given that relationship lending is an important mechanism for mitigating information frictions during the lending process (Fang, Ivashina and Lerner, 2013; Ivashina and Kovner, 2011; Sufi, 2007), the absence of ties implies that banks use other channels to access the requisite information.

This paper provides new evidence on one such channel – the bank syndication network – as a source of information during LBO loan origination. Understanding how a bank network functions and affects loan syndication and provision is an important, and relatively unexplored, question in finance. Existing work has focused mainly on network proxies of bank reputation and experience, such as its *Degree* and *Eigenvector*, to explain benefits in terms of reduction of information asymmetry and loan

¹ Typically, PE firms use debt to finance between 60–90% of the LBO (Gompers, Kaplan and Mukharlyamov, 2016; Kaplan and Strömberg, 2009). Global LBO deal volume was \$551 billion in 2019, with the US accounting for 41% of all LBO deals in that year (Bain and Company, 2020).

² Henceforth we collectively refer to PE firms and LBO targets as "borrowers" or "LBO borrowers". Our analysis shows that lead banks do not have pre-existing ties with the sponsoring PE firm (target) in 48% (79%) of the LBO deals in our sample. Similarly, non-lead banks in the syndicate do not have ties with the sponsoring PE firm (target) in 72% (89%) of the LBO deals in our sample.

raising (Godlewski and Sanditov, 2017; Godlewski, Sanditov and Burger-Helmchen, 2012). However, these and other traditional and widely used proxies are unsuited to capture the mechanisms of information propagation through the network.³

We exploit recent advances in social network research to make significant progress on the above challenges. Specifically, we employ two recently developed centrality measures, *Diffusion* and *Sourcing*, to capture a bank's capability to source information from the loan syndication network as well as to provide information to it. These measures were first introduced by Banerjee, Chandrasekhar, Duflo and Jackson (2013), and later refined by Banerjee, Chandrasekhar, Duflo and Jackson (2019), to understand information flows within village community networks.⁴ Our measures assume that information is passed stochastically from one node to another in the network with a fixed per-period transmission probability.⁵ *Diffusion* of a given bank measures the expected total number of times that any other bank in the network receives information broadcasted by this bank. *Sourcing* captures how often a given bank might expect to hear from every other bank in the network. Intuitively, *Diffusion* informs about the extent to which information spreads out from a given bank or, in other words, the bank's ability to send information within the network. *Sourcing*, on the other hand, informs about the extent to which a bank can be informed by other banks in the network. We refer to these jointly as *information centralities*.

Our analysis uses a sample of 2,414 LBO loans issued in the US between 1986 and 2012. This data is sourced mainly from Reuters' LPC DealScan, a database used by many previous studies on syndicated loans (Bharath, Dahiya, Saunders and Srinivasan, 2011; Demiroglu and James, 2010; Fang, Ivashina and Lerner, 2013; Fernando, May and Megginson, 2012; Ivashina, 2009; Ivashina and Kovner, 2011; Sufi, 2007).⁶

³ Godlewski, Sanditov and Burger-Helmchen (2012) and Godlewski and Sanditov (2017) use centrality measures based on network topology as proxies for bank reputation, and argue that better-connected banks have more reputation and are more informed, which helps them mitigate lending costs. However, the causal link between network topology and information access is unclear. By their very design, centralities based only on the network topology cannot inform about how information actually propagates through the network. Recent work in sociology indeed shows that reputed actors do not always diffuse more information into the network (Duarte, Finan, Larreguy and Schechter, 2019).

⁴ Banerjee, Chandrasekhar, Duflo and Jackson (2013) and Banerjee, Chandrasekhar, Duflo and Jackson (2019) denote these as *Diffusion* centrality and *Gossip* centrality, respectively. Although village networks are built on informal ties among residents, information flows in these networks are non-random and influenced by endogenous social and demographic factors like proximity, wealth status, occupation, religion, and caste (Dercon, 2005; Udry and Conley, 2004). While *Diffusion* and *Sourcing* were developed to study such *informal* village networks, the similarity of information dynamics in more formal settings provides a key motivation for this paper's use of *Diffusion* and *Sourcing* to study information flows within bank syndication networks.

⁵ This assumption is required because actual information flows between banks are not observable. However, Banerjee, Chandrasekhar, Duflo and Jackson (2019) show that the assumption of a fixed per-period information transmission probability does not affect the correct relative ranking of network nodes (banks) as a function of their *Diffusion* and *Sourcing* centralities in the network.

⁶ This paper focuses only on LBO loans rather than the entire corporate syndicated loan market for many reasons. First, the mean LBO loan is five times larger than the typical corporate loan, meaning that banks issuing LBO loans must contend with higher credit exposure and uncertainty. Second, over 91% of LBO targets are private firms that present significant adverse selection challenges to investing banks. This means that

We begin by showing that banks' information centralities are distinct from traditional centralities that are proxying bank reputation, like *Degree* and *Eigenvector* (Godlewski and Sanditov, 2017; Godlewski, Sanditov and Burger-Helmchen, 2012).⁷ Next, we estimate the effects of network centralities on the loan syndication process, which begins after the LBO borrowers appoint a *lead bank* to *arrange* the loan. The lead bank attempts to sell portions of the loan by inviting other banks and financial institutions within its network (Ivashina, 2009). Interested banks join the syndicate and contribute their agreed shares to the loan. Once the syndicate is formed and the required loan amount raised, the lead bank negotiates lending terms such as maturity, collateral, and interest rate with borrowers (Bruche, Malherbe and Meisenzahl, 2020). Figure 1 presents an overview of these steps.

[Figure 1]

Identifying how information flows within bank networks is empirically challenging for several reasons. First, networks are formed endogenously, and it is difficult to observe communication among banks and quantify how they affect loan syndication decisions. Second, LPC reports only completed loans, and does not provide data on banks that were invited to join a syndicate but did not eventually do so. This makes it difficult to observe banks' detailed actions and identify clear, unambiguous counterfactual decisions. Third, LBO loan syndication is unlikely to be random, and omitted factors could affect both bank-level information sharing and lending decisions, making it difficult to infer a causal relationship.

We address the above identification problems by developing suitable counterfactuals for realized syndication decisions using propensity score (PS) matching, wherein incumbent syndicate *members* are matched to ten other banks that did not actually join, but otherwise have similar characteristics. We refer to these collectively as the *candidate* banks. As participation in a syndicate is dichotomous for each of the candidate banks, we use conditional logit (CL) models to obtain consistent estimates from this matched sample and correct for unobservable heterogeneity at the deal level. To allay concerns of reverse causality, we use lags of centrality measures relative to the timing of a focal deal. This approach allows us to control for the endogeneity of potential differences in the LBO loan syndication process across less- and better-connected banks.

Controlling for prior ties and geographic distances between lead and candidate banks, and

gathering and sharing of information among banks, including through their networks, is crucial during syndication. Such issues are less likely in the case of syndicated loans issued for other purposes like capital expenditures or working capital. Third, LBOs involve transfer of ownership to the acquiring PE firm with the target undergoing significant reorganization post LBO (Bernstein and Sheen, 2016; Davis, Haltiwanger, Handley, Lipsius, Lerner and Miranda, 2019). The lending syndicate must evaluate the unique moral hazard challenges due to such changes (Gompers, Kaplan and Mukharlyamov, 2016), which do not exist for other loans. Lastly, PE firms are repeat players in the LBO market and tend to establish ties with banks (Ivashina and Kovner, 2011), which are less prevalent for borrowers of other loans. Focusing on the LBO market allows us to better identify whether the information spread via bank networks matters despite the presence of these ties.

⁷ The literature has also used other centrality measures, like for example *Betweenness* and *Katz-Bonacich* centralities. Given the high correlation between the latter and both the *Degree* and *Eigenvector*, and for the sake of conciseness, we only focus on the *Degree* and *Eigenvector* in the paper.

between candidate banks and borrowers, our CL estimates show that *Diffusion* and *Sourcing* have a substantial and positive impact on the odds that a candidate bank joins an LBO syndicate. Specifically, a one-standard-deviation increase in *Diffusion* (*Sourcing*) leads to a 41.5% (83.9%) rise in a candidate's odds of joining the syndicate. Thus, the more a candidate bank is expected to hear from the network (*Sourcing*), and the more it is plausibly informed, the greater is its probability to join the syndicate. Similarly, syndicates are more likely to include partners from which the lead banks have heard more frequently, as captured by the candidates' *Diffusion*.

If information that reaches banks through networks is so crucial during LBO syndication, a natural question is whether banks continue to rely on the network for information if they already know the borrowers or have worked with the syndicate banks in prior interactions. To answer this, we interact *Sourcing* with relationship proxies between candidate banks and borrowers. The corresponding CL estimates suggest that, when a candidate bank lacks prior ties to incumbent borrowers, information that could be sourced by a candidate bank from the network is instrumental to it joining the syndicate; the probability to join almost doubles as one moves from the 5th to 95th percentile for *Sourcing*. When a candidate bank has prior ties to incumbent borrowers, the probability of joining a syndicate also increases with the strength of *Sourcing*. However, the marginal effect of *Sourcing* on syndicate joining conditional on the candidate's ties to PE sponsors is stronger than its effect conditional on the candidate's ties to LBO targets. *Diffusion* exhibits similar behavior when it comes to syndicate joining. Taken together these analyses suggest that while having prior ties is always beneficial for the candidate banks' probability to join LBO syndicates, information centralities complement lending relationships in this setting.

Next, we investigate how the information centralities of the lead bank might affect its share of the LBO loan upon syndicate formation. This question is important since candidates will assess the need for ex-ante due diligence and ex-post monitoring of the borrower, and ascertain the lead bank's incentives to offer bad loans or shirk monitoring duties (Bharath, Dahiya, Saunders and Srinivasan, 2011; Ivashina, 2009; Sufi, 2007). Conditional on prior ties, if participating banks cannot acquire information relevant to the lead bank and borrowers from the network, they might try to minimize credit exposure and force the lead bank to hold a greater part of the loan. Similarly, the extent to which the lead bank can acquire information relevant to candidates and borrowers from the network might impact how much of the LBO loan it sells to incoming participants. Since our data does not allow us to capture actual information flows through the network, we interpret the observed lead bank share as an equilibrium outcome resulting from the overall information processed by lead and participant banks.

We use fractional response models to estimate the effects of network centralities on lead bank

share, which ranges between zero and one.⁸ Coefficients of lead bank *Diffusion* and *Sourcing* are negative and highly significant at the 1% level. A one-standard-deviation increase in lead bank *Diffusion* (*Sourcing*) leads to a 3.1% (2.6%) drop in the LBO loan share from the mean of 41.9%. This implies that lead banks that source (disseminate) more information from (into) the network manage to sell a greater share of the loan.⁹

Lastly, we assess whether information centralities impact LBO loan terms, comprising loan maturity, collateral requirements, and interest rate. We follow Melnik and Plaut (1986) and Dennis, Nandy and Sharpe (2000), and model LBO loans as *n*-dimensional contracts, where each dimension denotes a specific loan term that is set jointly with the other terms. We use a system of equations to model loan maturity and collateral jointly, and then model loan spread as being determined jointly by these two terms. To resolve the endogeneity problem associated with the joint determination of loan terms, we follow Bharath, Dahiya, Saunders and Srinivasan (2011) and use instrumental variables two-stage least squares (2SLS) regressions to analyze each equation.

Our findings suggest that a one-standard-deviation increase in lead bank *Diffusion* corresponds to a 0.11 standard deviation increase in loan maturity (equivalent to 1.2 months increase) for the mean LBO with a maturity of 66.8 months. Similarly, while half of the deals require collateral, a one-standard-deviation increase in lead bank *Diffusion* (*Sourcing*) lowers the probability of collateral demand by up to 16% (13%). Lastly, a one-standard-deviation increase in lead bank *Diffusion* (*Sourcing*) lowers the mean interest rate spread (over LIBOR) of 3.8% by 13 (11) basis points. In economic terms, the discounted cash saved from lower interest rates due to lead bank *Diffusion* is \$1.2 million over the mean loan period, implying a 0.6% higher return for every dollar invested by the PE firm. These results are consistent with the idea that information diffusion and sourcing capabilities by lead banks enable LBO loan issuance on cheaper terms.¹⁰

Our paper contributes to several strands of literature. First, recent studies explored the role social networks play in financial markets. For instance, Bajo, Chemmanur, Simonyan and Tehranian (2016) find that more central investment banks enjoy pricing advantages for their IPOs. Plagmann and Lutz (2019) show that more central VC firms attract higher quality peers as syndicate partners, while Hochberg, Ljungqvist and Lu (2007) show that funds of better networked VCs perform better. Li and

⁸ Since *lead bank share* is determined after the syndicate has been formed, we no longer require the PS matched sample, and instead use deal level data for the remaining analysis.

⁹ Besides controlling for bank-borrower ties and deal characteristics, we include cross-sectional reputation of the sponsoring PE firm as a proxy for target firm opacity. This proxy is included to account for the fact that reputation signals selection and monitoring skills of the PE firm (Demiroglu and James, 2010), and also because reputed PE firms tend to be more conservative and less risk-taking (Gompers, Kaplan and Mukharlyamov, 2016). LBOs by reputed PE firms should thus require less due diligence and be easier to syndicate. Our findings show that better reputation combined with a greater ability to source network information make it easier to assess and monitor the target. In such cases, lead banks find it easier to sell more of the LBO loan to syndicate members.

¹⁰ In comparison, coefficients of topological centralities are only significant up to the 10% level for all loan terms. This implies that network position is a weaker mechanism to resolve information asymmetries compared to information centralities.

Schürhoff (2019) show that more central bond dealers levy higher trading costs but provide faster execution. Two relevant studies on loan syndication by Godlewski, Sanditov and Burger-Helmchen (2012) and Godlewski and Sanditov (2017) use traditional centralities to argue that better-connected banks enjoy reputational advantages that help them mitigate lending costs. A common assumption in all these studies is that actors are fully aware of their network. We add to this literature by showing how banks learn from each other by sharing information via the network despite being unaware of its entire topology.

Second, our paper adds to the literature on syndication and partner selection decisions in financial markets. Studies show that investors syndicate to diversify their risks by co-investing (Ivashina and Scharfstein, 2010; Lerner, 1994). Ties built on repeat syndication create networks that facilitate both direct and indirect exchange of information among investors (Hochberg, Lindsey and Westerfield, 2015). "Central" investors within such networks have better knowledge of opportunities and competition (Hochberg, Ljungqvist and Lu, 2007; Sorenson and Stuart, 2001; Sufi, 2007), and have a reputation for being successful, all of which has a certification effect on the firms that they invest in (Ozmel, Robinson and Stuart, 2013; Robinson and Stuart, 2007). We complement this literature by showing that loan syndication depends on connections and information exchange among banks, without which they will be unable to diversify the risks present in their individual loan portfolios.

The rest of the paper is organized as follows. Section 2 describes the information and traditional centrality measures. Section 3 presents the sample and summary statistics. Section 4 describes the empirical design and results. Finally, Section 5 concludes.

2. Research framework

Our analysis focuses on the network of banks active in the syndicated US LBO loan market. Accordingly, we begin by describing the structure of this network. We then introduce network centrality measures that characterize the extent to which banks are expected to transmit information. These measures, denoted *Diffusion* and *Sourcing*, capture a bank's ability to diffuse and source information from its loan syndication network. In addition, we also consider two traditional *Centralities – Degree* and *Eigenvector –* in our analysis. Finally, we describe the type of information that banks plausibly exchange via this network.

2.1. Network structure

We determine the structure of the LBO loan syndication network based on the interactions between constituent banks. This structure is then used to estimate centralities that characterize each bank's relative position within the network. Our procedure is consistent with prior studies that have used social network analysis (SNA) techniques to understand venture capital (VC) syndication and performance (Hochberg, Ljungqvist and Lu, 2007; 2010), underwriting of initial public offerings (IPOs) (Bajo, Chemmanur, Simonyan and Tehranian, 2016), corporate finance policies (Fracassi,

2017), interbank lending (Kobayashi and Takaguchi, 2018), over-the-counter trading (Li and Schürhoff, 2019), and bilateral trade between countries (Richmond, 2019).

We define two banks *i* and *j* as having a tie (i.e., a prior interaction) in year *t* if they were part of at least one LBO loan syndicate in the past five years.¹¹ The collection of all such ties constitutes the prevailing syndication network, which can be represented as an *adjacency matrix*.¹² For example, an adjacency matrix representing the bank network for the year 2010 is computed using data on syndicated LBO loans issued between 2005 and 2009. Since the network evolves constantly due to changes in the loan syndication process, market trends, and the entry/exit of banks, we construct adjacency matrices on an annual rolling basis using five-year trailing windows.¹³ Our matrices are *undirected* (and thus symmetric), with each cell representing the number of past syndicates in which banks *i* and *j* were together. These adjacency matrices are then used to compute the relevant network centralities at the bank-year level. Banks that were not active in the market during a given five-year trailing period are considered "newcomers" if they were part of an LBO in the focal year, and consequently have their network centralities set to zero.

2.2. Centrality measures

2.2.1.Diffusion

Our first network measure is *Diffusion* centrality, which captures how information spreads from a given node through the network over a certain number of time periods. *Diffusion* is similar to contagion, and occurs because information generated at one node is passed on stochastically from neighbor to neighbor, along with details of the node that generated that information (Banerjee, Chandrasekhar, Duflo and Jackson, 2019). More formally, assume some information (e.g., pertaining to an LBO deal or to market conditions at the time of the deal) is initiated at bank *i*, and is broadcasted in the first period t = 1 among *i*'s neighbors with probability *p*. During each subsequent period, every informed neighbor shares *this information and the identity of the source* with its neighbors with the same probability *p*. This process is repeated over *T* time periods. The hearing matrix **H** is then defined as:

$$\mathbf{H}(\boldsymbol{g}_{y}, \boldsymbol{p}, T) = \sum_{t=1}^{T} (\boldsymbol{p}\boldsymbol{g}_{y})^{t}$$
(1)

¹¹ This approach is similar to the manner in which previous studies such as Hochberg, Ljungqvist and Lu (2007), Godlewski, Sanditov and Burger-Helmchen (2012), and Bajo, Chemmanur, Simonyan and Tehranian (2016) have constructed network-based centrality measures.

¹² Formally, a network of *N* distinct banks forms a N×N matrix $\mathbf{g} = f(i, j)$, where each element f(i, j) denotes a tie between banks *i* and *j*, and function *f* defines the weight of the tie.

¹³ To test the robustness of our definition, we re-ran all our analyses with ties measured over shorter time windows. Our results remain consistent to changes in the length of the trailing windows.

where g_y is the $N \times N$ undirected adjacency matrix of the bank network for a given year y. The *ij*-th element of **H** is the expected number of times that bank *j* hears some information that originated from bank *i* over *T* previous periods. Banerjee, Chandrasekhar, Duflo and Jackson (2013) show that the per-period transmission probability, *p*, can be reasonably well approximated by $1/E[\lambda_1(\mathbf{g})]$, which is the inverse of the largest eigenvalue of the adjacency matrix *g*. They also suggest that *T* can be approximated by the diameter of the network. *Diffusion* centrality is thus defined as:

$$Diffusion(\boldsymbol{g}_{y}, \mathbf{p}, \mathbf{T}) = \mathbf{H}(\boldsymbol{g}_{y}, \mathbf{p}, \mathbf{T}) \cdot \mathbf{1} = \sum_{t=1}^{T} (p\boldsymbol{g}_{y})^{t} \cdot \mathbf{1}$$
(2)

where **1** is a $N \times I$ column vector of ones. For a given year *y*, *Diffusion*_{*iy*} is the expected number of times that information originated by bank *i* is heard by any other member of the network over the time period *T*.¹⁴

The approximation of p by $1/\lambda_1(\mathbf{g})$ is an important design choice, as explained in Banerjee, Chandrasekhar, Duflo and Jackson (2019). If p is small ($\lambda_1(\mathbf{g}) > 1$), then very little information gets diffused over several time periods T, meaning that bank i transmits very little information onto the network. Hence its *Diffusion* is low. On the other hand, when p is large ($\lambda_1(\mathbf{g}) < 1$), information can spread quickly and saturate the network, hence bank i's *Diffusion* is high. Banerjee, Chandrasekhar, Duflo and Jackson (2019) recommend $p = 1/\lambda_1(\mathbf{g})$ as the optimal threshold at which information may diffuse to all nodes but does not get duplicated. A similar argument applies for using the network diameter $diam(\mathbf{g})$ as a proxy for T. When $T < diam(\mathbf{g})$, information from one node does not have enough time to reach all the other nodes. Alternatively, there is saturation if $T > diam(\mathbf{g})$ as information can reach some nodes multiple times as "echoes". Thus, Banerjee, Chandrasekhar, Duflo and Jackson (2019) recommend using $T = E[diam(\mathbf{g})]$ as information can diffuse but not necessarily duplicate across the network at this value.¹⁵

The interpretation of *Diffusion* in subsequent analyses is as follows. From a lead bank's perspective, higher (lower) *Diffusion* of a candidate bank implies that the former expects to receive more (less) information from the latter. Conversely, from a candidate bank's perspective, a lead bank with higher (lower) *Diffusion* is expected to share more (less) information. The nature of information that could be exchanged is discussed in Section 2.3 below. One would therefore expect a greater probability of syndication with banks having higher *Diffusion*.

¹⁴ Note that the dimension of **H** is the same as that of **g**. Since *p* is a scalar, and **g** is a matrix of $(N \times N)$, $\mathbf{w} = p\mathbf{g}$, is of $(N \times N)$. Any power of **w** is $(N \times N)$, hence the final matrix **H** is of $(N \times N)$. Multiplying **H** by **1** $(N \times 1)$ gives the column vector of $(N \times 1)$, where each row represents the *Diffusion* of the bank in the *i*th row.

¹⁵ The true values of p and T are unobservable to us. In unreported results, available upon request, we also run extensive simulations to understand the behavior of *Diffusion* and *Sourcing* further the changes in the underlying assumptions about p and T. Based on the arguments of Banerjee, Chandrasekhar, Duflo and Jackson (2013) and Banerjee, Chandrasekhar, Duflo and Jackson (2019), and our own simulations, we are confident that our inability to observe the true values of p and T does not inhibit the econometric conclusions of our main analyses.

2.2.2.Sourcing

Diffusion centrality considers information diffusion from the *sender's* standpoint. To consider how information reaches *receivers* in the network through this diffusion process, we once again follow Banerjee, Chandrasekhar, Duflo and Jackson (2019) and use the hearing matrix **H** to compute *Sourcing* centrality. *Sourcing* represents how often a given bank *j* "hears" information from the other banks over *T* periods. Recall that in each period, a bank receives information originating from various parts of the network. Since $\mathbf{H}(g_y, p, T)_{ij}$ is the expected number of times bank *j* hears information originating from bank *i*, the *j*-th column of **H** denotes bank *j*'s expected information sourcing from every other bank in the network. *Sourcing* centrality of bank *j* is then estimated as the average of the *j*th column of **H**. It is the mean expected number of times a bank hears (receives) information from the entire network. Higher *Sourcing* therefore implies that a bank is particularly well informed not only through its direct ties to other banks, but also through the indirect ties it has with other banks within the syndication network.

2.2.3.Degree

Degree centrality is the most widely used network measure and is the number of ties an actor has with other members of the network. The intuition behind *Degree* centrality, in the context of LBOs, is that better connected banks have more reputation than less connected peers. Given the adjacency matrix \mathbf{g} of ties among banks over a five-year trailing period, the *Degree* centrality of bank *i* in year *y* is defined as:

$$Degree_i = \sum_{j,i \neq j} x_{ij} \tag{6}$$

where *j* represents all banks excluding *i*. Thus, *Degree* is the sum of row (or column) *i* of the adjacency matrix **g**. As **g** is computed over a five-year trailing period, *Degree_i* is essentially the total number of interactions of bank *i* with LBO syndicate partners during the last five years.

Degree grows with network size as bigger networks have a greater pool of connected actors. This introduces a potential bias since bank networks evolve continuously, both in size and composition, making it difficult to use a single network or compare networks across time. Therefore, we normalize *Degree* by the maximum number of ties possible in an *n*-actor network (i.e., n-1).

2.2.4. Eigenvector

Although *Degree* captures ties between actors, it does not consider the quality of these ties. Specifically, *Degree* cannot distinguish between two focal banks in which one is connected to a group of banks that are not well-connected, while the other is connected to the same number of banks that are well-connected. While both banks have the same *Degree*, their influence within the network differs with the connectedness of their partner banks.

Consequently, banks having ties to better-connected banks will exert greater influence than those tied to weakly-connected banks. To capture these complexities, we use *Eigenvector* centrality.¹⁶ *Eigenvector* is a specialized variant of *Degree* centrality in which each tie of a given bank is weighted by the respective centrality of that connection. Thus, *Eigenvector* considers the number of connections as well as the centrality of each such connection and captures the extent to which the bank has ties to prominent and well-connected banks in the network. Formally, it is defined as:

$$Eigenvector_i = a \sum_{j=1}^{N} x_{ij} e_j \tag{4}$$

where a is a constant parameter representing the biggest eigenvector of the corresponding adjacency matrix, and e_j is the eigenvector centrality of bank j. To control for potential biases and ensure cross-comparability, we normalize this measure by the highest possible *Eigenvector* in a network of n actors.

2.3. Types of information travelling through the network

Although we do not directly observe the information flowing through the LBO loan syndication network, we can nonetheless provide some intuition on the kind of information exchanged over the network that would be instrumental during LBO loan syndication. First, banks may exchange information pertaining to the LBO deals currently under consideration. This can for example include private information on borrowers obtained from prior lending relationships or from other banks through the network. Second, lead banks may invite banks to join their LBO loan syndicate because the latter might share information on and to other banks that might be interested to join the syndicate. Third, interacting with informed banks could help lead banks keep track of latest developments in the syndicated loan market much more precisely compared to what they would be able to learn by simply accessing commercially available databases. This might include information on the buyout market regarding deal valuations, liquidity, loan terms and other conditions. Such information might directly impact loan terms and is otherwise hard to obtain from other sources.

3. Data and summary statistics

The primary data source on LBO loans is the Loan Pricing Corporation's (LPC) Dealscan database from Thomson Reuters. LPC data has been widely used in previous studies on loan syndication, particularly those related to LBOs (Bharath, Dahiya, Saunders and Srinivasan, 2011; Demiroglu and James, 2010; Fang, Ivashina and Lerner, 2013; Fernando, May and Megginson, 2012; Ivashina and Kovner, 2011; Sufi, 2007). We complement this transaction-level data with additional information on PE characteristics, such as age and funds raised, from Thomson One (formerly

¹⁶ Given that we use undirected (and thus) symmetric adjacency matrices, *Eigenvector* centrality is also equivalent to the Katz-Bonacich centrality (see Bonacich and Lloyd (2001) for details). We have verified this equivalence in unreported robustness checks.

Venture Economics). Information on the pre-LBO characteristics of target firms are available only if they were listed prior to the LBO and are obtained from Compustat. Loan data from LPC contains details of contributions made by lead arrangers and participants at the *tranche* level. Several such tranches, ranked in order of seniority, make up the overall debt package issued to finance an LBO.

Our initial sample consists of 65,362 syndicated loan tranches for 5,631 LBO transactions over the period 1986–2012. This data contains deal-specific information such as loan terms, identities of the participating banks, PE firms, and targets, and their observable characteristics. To eliminate heterogeneity of the institutional context, we restrict our sample to US-based LBO targets that were sponsored by US-based PE firms. However, we do not impose any such restrictions on banks and include LBOs arranged or funded by non-US banks. These restrictions result in a final sample of 2,414 LBOs comprising 5,766 individual loan tranches.¹⁷

3.1. Summary statistics

Table 2 presents summary statistics on the bank centralities.¹⁸ The mean *Diffusion* centrality of lead banks is nearly 2.5 times higher, on average, than that of members, suggesting that banks that are better at disseminating information may be preferred to form and lead loan syndicates. However, the *Sourcing* centrality of members is slightly higher than that of lead banks, with the *t*-test of the difference in their means statistically significant at 1% level. Lead banks also occupy more central network positions compared to syndicate members, as suggested by the higher *Degree* and *Eigenvector* centralities.¹⁹

Table 3 shows that *Diffusion* and *Sourcing* are only weakly correlated. This is expected since although both measures are derived from the same hearing matrix **H**, they represent different concepts. *Diffusion* tracks a bank's ability to *send* information into the syndication network whereas *Sourcing* tracks its ability to *receive* information from the same network. *Diffusion* is moderately correlated with *Degree* and *Eigenvector*, which is expected since diffusion centrality is proportional to *Degree* centrality for T = 1 and converges to eigenvector centrality when $\lambda_1(\mathbf{g}) > 1$ and $T \rightarrow \infty$ (Banerjee, Chandrasekhar, Duflo and Jackson, 2013; 2019). *Sourcing* is also partially correlated with *Eigenvector*, which shows that even with little intuition of the network structure, banks can still assess the correct topological ranking among members based on how often they hear about them over sufficient time periods. Lastly, *Degree* and *Eigenvector* are highly correlated with each other, implying considerable overlap between these traditional measures.

[Tables 2 & 3]

¹⁷ Note that a bank can participate in multiple loan tranches within an LBO debt package. However, we count each participating bank only once for an LBO.

¹⁸ All centralities are standardized in the empirical analyses to ensure cross-comparability.

¹⁹ See online Appendix A for pairwise correlations between *Diffusion* and *Sourcing* centralities of lead and syndicate member banks.

Summary statistics on the LBO sample are presented in Table 4. The mean (median) LBO loan package is about \$303 (\$207) million in size, has a maturity of 67 months and carries a spread of 3.8% above LIBOR, suggesting that LBO loans are substantially larger than standard business loans.²⁰ The median syndicate size, including lead banks, is four. One out of every two LBO loans is secured with collateral, suggesting the high level of perceived riskiness in these deals. Nearly seventy percent of deals were arranged by domestic banks.

Looking further at bank characteristics reveals that on average lead banks have stronger ties with PE firms than with targets prior to an LBO. Member bank ties with PE firms and targets exhibit a similar pattern but are marginally weaker than lead bank ties. The median bank has no ties with borrowers. In the case of lead-target and member-target relationships 75% of banks have no ties to LBO targets. These findings highlight that ties are virtually non-existent between banks and PE firms in over half the sample, and even more so among banks and targets.²¹

[Table 4]

The median age of the sponsoring PE firm is 17 years and only 8% of them were owned significantly by other financial institutions. Among publicly listed targets for which we have Compustat data, the mean firm size, measured in total assets, is \$1.48 billion and median pre-deal EBITDA margin is 13%. This suggests that the LBO targets in our sample (especially those publicly listed prior to their LBO) are large, mature companies which, unlike startups, require significant funding to be acquired.

3.2. Evolution of bank networks in the US LBO loan market

The US LBO loan market has evolved continuously over time. Figure 2 shows the annual number of banks active in this market between 1991 and 2012. The number of banks grew almost five-fold during the late-90s, possibly due to the rapid growth in buyout activity following the deregulation of private markets by the National Securities Markets Improvement Act (NSMIA) of 1996.²² This trend stalled after the Dot-Com bubble burst and then fell considerably after the 2007-09 financial crisis. Much of the activity in the syndicated LBO loan market is due to domestic banks.

²⁰ Daniels and Morgan (2010) provide evidence that LBO loans are larger in deal size, carry higher interest rates, and are significantly more levered than any other type of corporate syndicated loans.

²¹ Online Appendix B shows that syndicate characteristics and LBO lending terms are not very different when ties between banks and LBO borrowers are present or absent. While prior literature identifies relationship lending as an important mechanism for mitigating information asymmetries during the lending process (Fang, Ivashina and Lerner, 2013; Ivashina and Kovner, 2011; Sufi, 2007), the absence of such ties in the majority of LBOs implies that other mechanisms are being used for the resolution of information asymmetries. If lead banks have no private information available on borrowers, they may potentially rely on information produced and transmitted by other banks within the syndication network. Table 4 shows that this is indeed a possibility as syndicate members have more or less similar ties as lead banks to PE firms and targets. Member banks could therefore play a key role in facilitating LBO loan provision, particularly when the lead bank does not have sufficient ties with prospective LBO borrowers.

²² A recent study by Ewens and Farre-Mensa (2019) shows that NSMIA helped create large PE funds that led to a massive surge in demand for buyouts and growth equity investments.

[Figures 2 & 3]

Figure 3 provides six different snapshots of bank networks in the US LBO loan market between 1992 and 2012. Nodes in each graph represent banks that were actively syndicating LBO loans in the previous five years. Blue nodes denote domestic banks and red nodes depict foreign banks. Edges represent connections between two banks based on the intensity of their association in LBO loan syndicates during the preceding five years. Clearly, interactions among bank have intensified substantially in conjunction with the massive growth of the US LBO market in the past three decades. The LBO syndication network has become much more centralized over the years, characterized by a growing number of highly connected banks (visible at the core), both domestic and foreign.

Figure 4 plots the mean bank centralities per year across the sample period. Mean *Degree* – the mean number of nodes connected to each node – rose until the year 1999 but fell back ever since to its initial levels. The mean of the other centralities – *Diffusion*, *Sourcing*, and *Eigenvector* – have all fallen over time, consistent with a network characterized by growing interconnectedness and an emerging core-periphery structure where some nodes are better connected than others. This is evident in the skewness of the distributions of bank centralities over time, as seen in Figure 5.

[Figures 4 & 5]

All four centralities have a moderately positive skew, implying the existence of a few highly connected and many sparsely connected banks. The skewness of *Diffusion* peaked in year 2000 but has dropped ever since. This signals that banks in general have become better over time at diffusing information to one another across the syndication network. Skewness of the other three centralities – *Sourcing, Degree,* and *Eigenvector* – increased until 2008, but has dropped ever since, which is also consistent with growing interconnectedness.²³

3.3. Relationship between bank network centralities and reputation

The syndicated LBO loan market is greatly affected by the information asymmetries between lead banks, syndicate members, and borrowers (Ivashina, 2009). These frictions must be alleviated for LBO syndicates to be formed, for deals to be financed, and for the loan market to clear.

In the case of syndicate formation, prior studies note that prospective members rely on reputation of the lead bank in deciding whether to join the syndicate (Carey and Nini, 2007; Ross, 2010; Sufi, 2007). The premise here is that reputation is valuable and serves as a credible signal of the bank's ability to conduct due diligence and screen and monitor borrowers (Dennis and Mullineaux, 2000). In addition, reputation is also beneficial for resolving information asymmetry present within the syndicate (Bharath, Dahiya, Saunders and Srinivasan, 2011; Diamond, 1984; 1989; 1991; Ivashina, 2009). Relatedly, papers on loan syndication by Godlewski, Sanditov and Burger-Helmchen (2012)

²³ Online Appendix C provides a visual illustration of the market share of individual banks in the US LBO market by year.

and Godlewski and Sanditov (2017) use traditional centralities such as *Degree* and *Eigenvector* as proxies for bank reputation. These studies argue that more central banks attract more attention from other banks in the network, and thus enjoy a reputational advantage that helps them mitigate lending costs.

However, these arguments make the default assumption that more reputed banks (that are topologically more central by design) have better access to information flowing within the network.²⁴ By their very design, traditional centralities and other proxies of reputation cannot capture information flows (Banerjee, Chandrasekhar, Duflo and Jackson, 2013; 2019). Being topologically central also does not automatically imply that banks will actively disseminate or seek information from the network. Thus, better constructs that can capture the expected information exchange among banks are required. This is where information centralities become prominent as they encompass an actor's probability to pass information along the network together with the identity of its source (see Sections 2.3.1 and 2.3.2). As such, they do not require implicit assumptions linking reputation to the extent to which banks possess and exchange information.

A natural question then is the extent to which information centralities and traditional centralities differ from each other. While *Diffusion* and *Sourcing* are constructed on the premise that even naïve actors can develop accurate knowledge by simply hearing about each other, it could well be that these centralities merely facilitate the identification of dominant banks in the network. If this is true, then information centralities would be similar to traditional centralities and act as mere proxies for bank reputation. The alternative hypothesis is that information centralities capture how banks interact with each other and share information, and thus represent something different than reputation.

We assess this empirically by modeling the network centrality of a bank as a function of its market share (which serves as an additional proxy for its reputation in the LBO loan market). Formally, we use the following specification:

$$Centrality_{it} = \beta_0 + \beta_1 Bank \ LBO \ Market \ Share_{it} + \alpha_i + \mu_t + \varepsilon_{it}$$
(5)

where *Centrality_{it}* denotes any of our four centrality measures of bank *i* in year *t*, *Bank LBO Market Share_{it}* is the proportion of LBO loan syndicates (in dollar terms) in which bank *i* participated during the five years preceding year *t* (see Table 1 for definitions), α_i and μ_t are vectors of bank and year fixed effects, respectively, and ε_{it} is the error term.

[Table 5]

The results presented in Table 5 clearly show that, controlling for bank and time fixed effects, bank market share is not correlated with *Diffusion*. This is consistent with the idea that banks with

²⁴ Bajo, Chemmanur, Simonyan and Tehranian (2016) make similar assumptions when studying IPO underwriter networks. Hochberg, Ljungqvist and Lu (2007) do the same to examine VC networks.

bigger market shares do not systematically push information into the network, possibly to maintain their dominance and prevent other banks from competing effectively against them. Market share is also not correlated significantly with *Sourcing*, suggesting that reputation, by itself, does not guarantee the bank's ability to acquire information from other banks. In contrast, coefficients of *Degree* and *Eigenvector* are consistently positive and statistically significant, meaning that banks with higher market shares – or reputed banks – occupy topologically central positions within the LBO syndication network. Together with the correlation estimates in Table 3, these findings imply that the information and traditional centralities we employ are quite distinct from one another. We hence expect these measures to behave differently our further analyses of syndicate formation and deal terms.

4. Empirical design and results

Our empirical approach is based on the stylized process through which LBO loans are syndicated, as depicted in Figure 1. We make the simplifying assumption that the lead bank is already chosen to arrange financing for the LBO through syndication. The analysis is organized in two stages. First, we investigate whether the network centralities of a candidate bank influence its participation in an LBO loan syndicate and the share retained by lead bank(s) in the syndicate. Second, conditional on successful syndicate formation, we analyze the relationship between lead bank centralities and the LBO lending terms.

4.1. Network centralities and syndicate structure

4.1.1.Syndicate participation

We begin by asking whether the probability of a candidate bank's participation in an LBO loan syndicate is influenced by its abilities to share (receive) information with (from) the network.

To perform this analysis, we would ideally need data on banks which joined LBO syndicates as members and those that were invited but did not join. Unfortunately, LPC reports only completed LBO loan syndicates and does not provide data on banks that opted out of joining or were not invited to join. To overcome this limitation and develop requisite counterfactuals for analysis, we could simply consider every bank that participated in an LBO during the five preceding years as a candidate for the given syndicate.²⁵ For each deal, we could then stack all incumbent members and candidate banks to create a *case-control* sample of LBO syndicate participations.²⁶ However, this approach implies that the mean lead bank would have to choose among 324 candidates to form an LBO syndicate comprising on average 2 or 3 members. This is unrealistic as attempting to market the loan

 $^{^{25}}$ This is also consistent with the rolling time window we use to construct networks centralities and other variables (see Section 2.1).

²⁶ A case-control sample of banks created in this manner contains 751,977 observations.

to so many candidates and sharing information with each of them is implausible from a cost and time perspective.

A more realistic assumption is that lead banks would first approach candidates that are closest to them in activity and characteristics within the syndication network. We therefore construct a casecontrol sample of LBO syndicate participation using the propensity score (PS) matching method (Dehejia and Wahba, 2002; Rosenbaum and Rubin, 1983).

In particular, we first identify all banks that were active in the LBO market during the five-year trailing period relative to the focal LBO. From these, we identify candidate banks that are closest to the incumbent members along three observable characteristics: prior experience arranging LBO loan syndicates (*Bank Lead Experience*), bank market share (*Bank LBO Market Share*), and domicile (*Non US Bank*) – all determined at the time of the LBO.²⁷ We then employ PS matching (Dehejia and Wahba, 2002; Rosenbaum and Rubin, 1983) and estimate scalar distance between the vectors of matching characteristics of the incumbent members and each candidate bank. Next, we sort the candidate banks based on this distance and select the ten nearest neighbors. Thus, incumbent members in a deal (*cases*) are matched to ten nearest neighbors (*controls*). All our matching is done with replacement (so that candidates can be used for matching more than once) as it reduces bias in the estimates (Abadie and Imbens, 2002).

$$Pr(Match_{ij} = 1) = \beta_0 + \beta_1 \cdot Cand \ Centrality_i + Controls'_{ij} \cdot \beta_2 + \mu_j + \alpha_t + \varepsilon_{ij}$$

$$6)$$

Our baseline specification has the following generic form:

where *Match_{ij}* is equal to one if bank *i* is member of the syndicate for LBO deal *j*, and zero otherwise. *Cand Centrality_i* denotes any of our four centrality measures for candidate bank *i*. *Controls_{ij}* is a set of controls capturing observable characteristics such as prior relationships between bank *i* and the incumbent borrowers that could influence syndicate joining decisions. μ_j and α_t denote deal and time fixed effects while ε_{ij} is the error term.

While the PS matching methodology allays potential endogeneity concerns, some unobservable heterogeneity could still bias our results. To resolve this issue and achieve better identification, we use the geographic distance between candidate banks and PE firms (*Bank–PE Geodist*) and between candidate and lead banks (*Bank–Bank Geodist*) as exogenous regressors. The logic stems from prior literature that banks incur distance-related screening and monitoring costs, and prefer to engage with borrowers and other banks in their vicinity (Degryse and Ongena, 2005). The relevance also stems from Sufi (2007) who notes that whenever problems of information asymmetry are more severe, lead

²⁷ Our PS matching strategy does not account for prior interactions between banks. The reason is that we use these variables as controls in our main empirical analyses. See online Appendix D for a detailed description of our propensity-score matching procedure.

banks are likely to choose members that are geographically closer to the borrower. Hence, the key identifying assumption is that geographic distance impacts syndicate formation, while also indirectly affecting LBO lending terms.²⁸

Given the dichotomous nature of our dependent variable and deal level matching of the sample, we employ conditional logit (CL) models (McFadden, 1984) to obtain consistent parameter estimates.²⁹ A major advantage of CL models is that they correctly account for unobservable heterogeneity at the deal level. All such systematic differences are fixed at the deal level, and do not affect the *within-deal* odds of one bank becoming a syndicate member over another. Examples of such fixed deal-level heterogeneity include differences among borrowers and their ties to lead banks.

Our main goal is to understand whether information centralities impact the probability of candidates becoming syndicate members. If banks are better able to, among other things, diffuse and source information from the network, we can expect positive effects of these centralities on the probability to become a syndicate member. However, prior literature states that information can also be gathered through prior lending relationships. It is therefore important to account for these in our models. Following Bharath, Dahiya, Saunders and Srinivasan (2011) and Ivashina and Kovner (2011), we control for prior ties between candidate banks and PE firms (*Bank–PE Relationship*), between candidate banks and targets (*Bank–Target Relationship*), and between candidate and lead banks (*Bank–Bank Relationship*).³⁰

[Table 6]

Table 6 reports the CL estimates of model (2). Models 1–3 report exponentiated coefficients with robust standard errors. Model 1 suggests that for a candidate bank the odds of joining an LBO syndicate increase by 83.9% (41.5%) for a one-standard-deviation increase in *Sourcing (Diffusion)* (both significant at 1% level). The size of the coefficient of *Sourcing* suggests that the ability to source information from the network plays a key role. Similarly, the coefficient of *Diffusion* implies that lead banks might also benefit from any relevant information relayed by candidates. This suggests that controlling for a candidate bank's access to information (via *Sourcing*), prior relationships with borrowers, and geographical proximity to both lead bank and borrowers (all of which are significant), its ability to share information (as captured by *Diffusion*) further enhances its probability of joining the LBO syndicate.

In Models 2 and 3 we test the effects of the traditional centralities *Degree* and *Eigenvector* on syndicate joining (we are unable to include them simultaneously as they are highly correlated). Both

²⁸ In unreported analyses, we include the geographic distance between candidate banks and LBO targets, which were found to be insignificant. This leads to the conclusion that geographic distance between banks and targets may not be a determining factor in the provision of LBO loans.

²⁹ These are sometimes also referred to as fixed-effect logit models.

³⁰ In robustness checks, we also measured relationships as the number of interactions in the preceding five years between PE firms and banks, and between targets and banks, respectively. Our findings are robust to these alternatives.

centralities have large, positive, and statistically significant effects on the odds of becoming a syndicate member. This is in line with the findings in Table 5, implying that reputation, proxied by the traditional centralities, also plays an important role in syndicate formation.

Lastly, we introduce all the variables in Model 4 and conduct a post-LASSO analysis to pick out the variables most strongly associated with the outcome, *Match_{ij}*. Post-LASSO is a two-step procedure from Belloni and Chernozhukov (2013) in which the first step uses the LASSO method to select the variables that best predict the outcome and the second step applies standard OLS to regress the outcome on these chosen variables.³¹ We find that *Diffusion* loses its significance, *Degree* disappears, while both *Sourcing* and *Eigenvector* emerge as strong predictors of syndicate joining (significant at 1% level). A one-standard-deviation increase in *Sourcing* and *Eigenvector* is associated with a 5.3% and 7.1% increase in the probability of joining the syndicate, respectively. This shows that successful syndicate participation depends most on a candidate bank's ability to source information from the network, and also its position within the syndication network. The findings are consistent with the general notion that the syndication network is useful for alleviating potential information problems arising during loan syndication.

A concern with *Diffusion* is that this measure is agnostic of a given syndicate; it captures the ability of a bank to send information to the entire network and not to just any particular lead bank of a focal syndicate. In additional tests, we therefore introduced a more restrictive, deal-specific diffusion measure, which captures the information sharing between the lead and a candidate bank. *Diffusion*_{ij} thus represents the expected number of times a lead bank *j* (*j*-th column of the hearing matrix \mathbf{H}_y , with y = 1991, ..., 2012) received information from bank *i* (i-th row of \mathbf{H}_y), evaluated at a time when *i* is a candidate bank for the focal syndicate of *j*. We run two tests. First, we estimated the correlation between *Diffusion*_{ij} and the original *Diffusion* wariable with the new *Diffusion*_{ij} and estimated the model (1) of Table 6. The results were very consistent, and the point estimates are of similar magnitude in both cases. Our findings are hence robust to this alternative specification of *Diffusion*.³²

4.1.2. Do networks complement or substitute for relationships?

If bank networks are an important source for enhancing information flows during loan syndication, a natural question arises whether information from the network is as important when banks are aware of the borrowers through prior interactions. This question is important because a prior relationship with the borrower may substantially reduce a bank's dependence on the network for

³¹ An advantage with a penalized method like LASSO is that coefficients that contribute most to an increase in the squared sum of errors are shrunk to zero. Belloni and Chernozhukov (2013), Belloni, Chernozhukov and Hansen (2014), and Belloni, Chernozhukov and Hansen (2014) show that running OLS on the variables chosen by LASSO (in the first step) provides consistent estimates.

 $^{^{32}}$ Given that both these measures yield identical results in regression analyses, we do not present these findings in the paper. These results are, however, available on request. For the same reason, we stick to the original definition of *Diffusion* proposed by Banerjee, Chandrasekhar, Duflo and Jackson (2013).

information relevant to the LBO. Under this logic, information centralities and borrower relationships should serve as substitutes. It must be noted, however, that bank-borrower relationships are virtually non-existent in more than half of the deals in our sample (see corresponding median for member-target relationships in Table 4). The scarcity of these ties leads us to believe that the syndication network may serve as an important source of information during syndicate formation.

We investigate these effects by interacting network centralities and borrower ties of candidate banks and report the CL estimates in Table 7. Models 1 and 2 show the interaction effects between information centralities and *Bank-PE Relationship* and *Bank-Target Relationship*, respectively. We repeat this exercise by interacting the traditional centralities with *Bank-PE Relationship* and *Bank-Target Relationship* and *Bank-Target Relationship* and *Bank-Target Relationship* and *Bank-Target Relationship* in models 3 and 4, respectively. Model 5 shows post-LASSO OLS estimates by pooling all the variables of interest and their interaction terms into a single specification.

[Table 7 & Figure 6]

The coefficients of centralities in Models 1–4 are consistent with the results in Table 6. Coefficients of the interaction terms between *Diffusion* and *Bank-PE Relationship* and *Bank-Target Relationship* in Model 1 are both significant (at 1% level) and slightly below one, implying their negative influence on the odds of joining a syndicate. In Model 2, the interaction between *Sourcing* and *Bank-PE Relationship* is significant (at 1% level) and above one, implying a positive effect on the bank's odds of joining a syndicate. In Models 3 and 4, the interaction terms between a candidate's traditional centralities and borrower ties are significant (at 1% level) and negatively associated with the odds of joining a syndicate. The positive association between *Sourcing* and *Bank-PE Relationship* persists in the post-LASSO results in Model 5. Similarly, the negative interaction effect of *Eigenvector* and *Bank-Target Relationship* also prevails in the post-LASSO estimation.

Since we use CL models to estimate the odds of a bank becoming a syndicate member, the interaction terms cannot be interpreted directly from the regression output. To better interpret the results, we present the corresponding interaction plots in Figure 6. Each plot shows, *ceteris paribus*, the effect of a network centrality over a discretized version of borrower relationship (indicating whether or not prior ties with the borrower exist) on a bank's probability of joining an LBO syndicate.

Plots (a) and (b) show the interaction effects between *Diffusion* and *Bank-PE Relationship* and *Bank-Target Relationship*, respectively, while plots (c) and (d) show the interaction effects between *Sourcing* and *Bank-PE Relationship* and *Bank-Target Relationship*, respectively. The *black* line denotes the situation where candidate banks have no prior ties with incumbent borrowers and the *blue* line denotes the existence of prior ties. Plots (a) and (c) suggest that a candidate bank always has a higher probability to join the syndicate if it has a prior relationship with the PE, irrespective of its *Diffusion* and *Sourcing* centralities. At the same time, the probability of joining the syndicate also increases proportionally with *Diffusion* or *Sourcing* regardless of prior ties. This is consistent with the idea that Bank-PE relationships and information centralities complement each other, i.e., information

that is flowing through the bank network is relevant and complementary to information that is gained through prior interactions with the PE.

The situation is slightly different when considering prior ties to the target (plots (b) and (d) respectively). Again, having prior ties to the target increases the probabilities of joining the syndicate. When a bank has no prior ties to the target, both its *Diffusion* and *Sourcing* centralities matter: both significantly increase the probability of joining the syndicate. Specifically, when banks lack ties to the target, their probability of syndicate joining is significantly lower if *Diffusion* or *Sourcing* is low (5th percentile) than when their *Diffusion* or *Sourcing* is median or high (95th percentile). However, when the bank has prior ties to the target, the increase in probability of joining the syndicate increases only slightly for better networked banks, implying that the information gained through the network as reflected by *Diffusion* and *Sourcing* centralities is only slightly complementary to information gained through direct interaction with the target (as the blue lines are relatively flat). Hence, information on the target through previous interactions is highly important, but less than 7% of the candidate banks do have a previous tie with the target company. When no such tie exists, having higher information centralities is important.

The interaction plots for *Degree* and *Eigenvector* are nearly identical (plots (e)–(h)). Even when *Degree* or *Eigenvector* is low, prior ties make it significantly easier to evaluate borrowers and improve a candidate's probability to join the syndicate. The utility of such ties diminishes with greater *Degree* or *Eigenvector* to the extent that their effect is indistinguishable from that of relationships at the right tail of the *Degree* or *Eigenvector* distribution. This suggests that having prior ties to either the PE or the target completely substitutes for having a strong *Degree* or *Eigenvector* network position (95th percentile), and hence that the network position does not bring in complementary information.

Overall, the results suggest that information centralities remain important regardless of whether a bank has prior relationships with borrowers. This implies that lead banks generally benefit from the information sourced and shared by candidates.³³ These findings are also consistent with the idea that the syndication network facilitates access to other types of relevant information than just about the PE and target firm associated with the focal LBO deal.

4.1.3. Network Centralities and Lead Bank Share

Prior literature states that in the presence of information asymmetries syndicate members require lead banks to hold a greater share of the loan (Bharath, Dahiya, Saunders and Srinivasan, 2011; Ivashina, 2009; Sufi, 2007). Understanding the relationship between lead bank share and the cross-

 $^{^{33}}$ *Diffusion* and *Sourcing* may in fact complement each other. To confirm this hypothesis, we run a regression identical to model (1) in Table 6 with a relevant interaction term. The exponentiated coefficient of this interaction term is 1.647, and is significant at 1% level, which confirms our hypothesis.

sectional variation in leads banks' information centralities is therefore important. The baseline

Lead Bank Share_j =
$$\beta_0 + \beta_1 \cdot Lead \ Centrality_j + \beta_2 \cdot Opaque + \beta_3 \cdot Lead \ Centrality_j * Opaque + Controls'_j \cdot \beta_4 + \gamma_i + \alpha_t + \varepsilon_j$$
7)

(

specification we use to investigate this is outlined as follows:

where the variable of interest is the percentage of the LBO loan retained by the lead bank(s) relative to the syndicate members (*Lead Bank Share*). Since lead bank share is determined after the syndicate is formed, we no longer require the case-control sample or candidate bank centralities used thus far, and instead use deal-level data for the remaining analysis. For each deal *j*, *Lead Centrality* denotes any of our four centrality measures for the lead bank. *Controls_j* is a set of control variables including prior ties between the lead bank/members and borrowers, ties between lead banks and members, and a variety of controls for deal characteristics. γ_i and α_t denote two-digit SIC target industry and time fixed effects while ε_j is the residual error term.

Our main coefficient of interest β_1 shows how the lead bank's network centrality influences its share in the LBO loan. We include Opaque proxying for the extent to which a lending syndicate must investigate and monitor the target. This variable allows us to differentiate among LBO targets based on their information opacity. The coefficient β_2 indicates how lead bank share changes with the target's perceived "opacity". The easiest way to measure target opacity is to look at whether the firm was listed on a stock exchange prior to its LBO, and follows from prior studies that regard public firms as being more transparent than private counterparts (Saunders and Steffen, 2011; Saunders, Steffen, Freudenberg and Imbierowicz, 2012). However, this approach is infeasible in our case as only 7% of targets in the sample were public firms prior to their LBO. To resolve this problem, we use reputation of the sponsoring PE firm as a proxy for target firm opacity. The choice of this proxy stems from two stylized facts. First, the reputation of a PE firm provides information about its LBO target selection and monitoring capabilities, and affects lenders' perceptions on riskiness of the LBO (Demiroglu and James, 2010; Ivashina and Kovner, 2011). Second, PE firms tend to become more conservative and less risk-taking as their reputation grows (Gompers, Kaplan and Mukharlyamov, 2016; Ljungqvist, Richardson and Wolfenzon, 2020). Following these arguments, we define our proxy, *PE Reputation*, as the total amount of funds raised by the PE firm in the five years preceding the LBO.

As the outcome variable *Lead Bank Share* is a fraction, we follow Papke and Wooldridge (1996) and use fractional response regressions to analyze equation (3). For LBO loans arranged by more than one lead bank, we take the within-deal averages of lead bank centralities and other characteristics. Finally, the standard errors are robust to heteroskedasticity and clustered at the PE firm level.

[Table 8]

The regression estimates are presented in Table 8. The top two rows show that greater diffusion and sourcing capabilities of lead banks result in lower stakes retained by them in the LBO loans. Marginal effects analysis on model (1) shows that a one-standard-deviation increase in *Diffusion* (*Sourcing*) is associated with a 3.1% (2.6%) reduction in the loan share retained by the lead bank. The interpretation is straightforward. Lead banks that have high information centralities are likely to hear and diffuse more information about the target, about other banks, and about the state of the LBO market, thereby reducing information asymmetries. As a result, syndicate members agree that the lead bank holds a smaller share of the LBO loan.

4.2. Network centralities and LBO borrowing terms

After investigating the impact of bank networks on syndicate formation, we now turn our attention to how they affect LBO loan terms. The observable loan characteristics we consider are loan maturity, collateral requirements, and loan interest rates. As outlined in Figure 1, these terms are set in the final stage of the loan syndication process during deliberations between the lending syndicate and borrowers, with the lead bank serving as an intermediary between them (Bruche, Malherbe and Meisenzahl, 2020; Ivashina, 2009). Our main focus here is to understand the information-related roles played by the lead bank during this final phase of loan syndication, as proxied by its *Diffusion* and *Sourcing* centrality. We hereby focus on the lead bank only, as this bank plays an instrumental role in setting the deal terms.

A potential concern is that LBO contract terms are jointly determined during loan syndication and hence cannot be analyzed independently of each other. This problem has been highlighted by Melnik and Plaut (1986) and Dennis, Nandy and Sharpe (2000) who model loans as n-dimensional packages in which each dimension represents a specific loan term such as spread, maturity or collateral that cannot be split or traded separately. Banks offer a bundle of these n-dimensional packages to borrowers, allowing them to tradeoff various terms in determining their optimal choice of loan package. This approach suggests that the contractual terms of a loan could be interrelated to each other. While some studies have examined loans (including LBO loans) as multi-dimensional contracts (Bae and Goyal, 2009; Benmelech, Garmaise and Moskowitz, 2005; Bharath, Dahiya, Saunders and Srinivasan, 2011; Dennis, Nandy and Sharpe, 2000; Graham, Li and Qiu, 2008; Qian and Strahan, 2007), none to the best of our knowledge have investigated the effects of bank networks on loan contract terms. Following Dennis, Nandy and Sharpe (2000) and Bharath, Dahiya, Saunders and Srinivasan (2011), we model loan maturity and collateral jointly and thereafter model loan spread as being determined by the choice of maturity and collateral. We express these choices mathematically using the following system of equations:

 $Maturity = \beta_{10} + \beta_{11}Centrality + \beta_{12}Collateral + Controls'_k\beta_{13} + X'_1\beta_{14} + \varepsilon_1$ 8.1)

$$Collateral = \beta_{20} + \beta_{21}Centrality + \beta_{22}Maturity + Controls'_k\beta_{23} + X'_2\beta_{24} + \varepsilon_2$$
8.2)

where *Maturity* is the duration of the LBO loan, measured in log months. *Collateral* is a dummy variable equal to one if the target provided some collateral as security against the loan amount, and zero otherwise. *Spread* is the interest rate charged on the loan (including fees) in excess of the prevailing "London interbank offered rate" (LIBOR).

(

To resolve the potential endogeneity associated with the co-determination of loan contract terms, we use instrumental variables two-stage least squares (2SLS) regressions to analyze the above system of equations. X_i represents exogenous instruments that identify each equation. Our choice of instruments is based on the prior literature on loan syndication and bank lending.

For Equation (4.1), we follow Brick and Ravid (1985) and use *Term Spread* as a source of exogenous variation for LBO loan maturity. Term spread is the difference in yields on ten-year and one-year US government bonds at the time of LBO loan issue, as reported by the US Federal Reserve. Brick and Ravid (1985) show that when the yield curve is upward sloping, using longer-term debt is preferred as it increases the present value of tax benefits. Conversely, shorter-term debt is preferred under a decreasing term structure. Since LBOs provide tax shields that allow for the interest paid on debt to be deducted as expense, we expect a positive relationship between LBO loan maturity and term spreads.

For Equation (4.2), we use *Syndicate Herfindahl* representing the degree of loan concentration within the syndicate as an instrument for collateral. It is measured as the sum of the squared individual shares of each bank (including lead banks) in the LBO syndicate. This instrument choice is due to Sufi (2007), who states that collateral is demanded when the information asymmetry between borrowers and the lending syndicate is high. Concentrated syndicates are able to monitor their borrowers more intensely and hence are expected to demand less collateral.

Lastly, for Equation (4.3), we use the *Average 6m Spread* charged by banks on all LBO loans issued in the preceding six months as a source of exogenous variation for loan spreads. Bharath, Dahiya, Saunders and Srinivasan (2011) note that this measure denotes recent trends in the interest rates being charged on syndicated loans and is widely used by banks as a benchmark in pricing new loan issues. DealScan provides this data in the form of pricing grids in which average spreads are specified according to borrower industry, size, and rating. It is therefore plausible to argue that the lagged average 6-month spread has a direct influence on the interest rate spread levied on an LBO loan; however, it is unlikely to have any direct influence on the non-price terms such as loan maturity and collateral.

A final problem in estimating our system of equations is that while spread and maturity are continuous variables, collateral is a binary. We address this issue by using a two-stage estimation in line with Dennis, Nandy and Sharpe (2000) and Bharath, Dahiya, Saunders and Srinivasan (2011). In the first stage, we use logit models to estimate *Collateral* as a function of all exogenous factors and then use the predicted values from these estimates as additional instruments for collateral in the 2SLS regressions. Bharath, Dahiya, Saunders and Srinivasan (2011) note that this correction provides consistent results of the 2SLS estimation.

The results of the estimations are presented in Table 9. All the regressions control for target industry and time fixed effects. To maintain clarity, we discuss the regression results of each equation separately.

[Table 9]

4.2.1.Maturity

Panel A reports the 2SLS estimates for *Maturity*. The coefficient of *Diffusion* is positive and statistically significant at the 1% level. A one-standard-deviation increase in lead bank *Diffusion* corresponds to 0.11 standard deviations rise in LBO loan maturity. Consequently, syndicate members can determine their monitoring needs more precisely and are hence willing to issue the loan for longer durations. Information flows in the opposite direction, i.e. from the network to the lead bank, might also be relevant, but do not seem to impact loan maturity as shown by the lack of significance for *Sourcing*. None of the other centralities, *Degree* and *Eigenvector*, is significant. This is in line with our previous findings that information centralities take precedence over traditional centralities in the context of LBO financing.

We also estimate the economic significance of the impact of bank network centralities on LBO loan maturity. Based on the coefficients reported in Panel A of Table 9, LBO borrowers obtain loans for a 1.2 months longer duration for a one standard deviation increase in lead bank *Diffusion*. The mean cash savings from this longer duration, when discounted over the holding period (5.6 years) at a mean rate of 5.09% (12-month LIBOR of 1.25% + average LBO loan spread of 3.84%) amounts to \$0.1 million. This translates to a 0.05% increase in return on investment for an initial equity contribution of \$198 million by the PE firm. Further details of these economic effects (and the assumptions made to estimate these effects) are available in the online Appendix E.

Loan maturity also increases with *Term Spread* as predicted by Brick and Ravid (1985). Finally, maturity also increases with target firm profitability (measured as the EBITDA-to-assets ratio).

We conduct several diagnostics tests to verify the validity and relevance of our instrument for *Collateral*. We conduct Durbin-Wu-Hausman χ^2 tests to determine whether *Collateral* is endogenously determined with *Maturity*. The p-values of these tests are all above 0.1, suggesting that these terms are set independent of each other. To test for correlation between *Collateral* and its instrument we report the F-statistic from the first stage. The lowest F-statistic value is 14.746, which

is above the minimum recommended value of 10 (Staiger and Stock, 1997), thus rejecting the null hypothesis of a weak instrument.

4.2.2. Collateral requirements

Collateral plays an important role in lending as it provides banks with a certain claim against the target's assets in the event of default and reduces the riskiness of debt (Berger and Udell, 1990). It thus helps mitigate problems of adverse selection and moral hazard between borrowers and lenders (Berger, Espinosa-Vega, Frame and Miller, 2011; Bharath, Dahiya, Saunders and Srinivasan, 2011). However, demand for collateral may be less stringent if the lead bank can leverage its network to resolve these asymmetric information problems.

Panel B shows how lead banks' network attributes affect the probability that a target will be asked to pledge collateral against its LBO loan. The coefficients of both *Diffusion* and *Sourcing* are significant at 1% level and negative in model (2). A one standard deviation increase in *Diffusion* (*Sourcing*) reduces the probability of collateral demand from 50% (average) by up to 16% (13%).³⁴

Degree and *Eigenvector* are also negative (models (3)–(4)), but the statistical significance is only 10%. Lastly, the post-LASSO estimates in model (5) suggest that only *Diffusion* has a stable negative effect on the probability of imposing collateral. These results suggest that resolution of information problems within the syndicate due to the network diffusion and sourcing capabilities of the lead bank reduces collateral requirements. They also confirm once again that traditional centralities of lead banks are a weaker mechanism to deal with information asymmetry within the syndicate compared to information centralities.

We also find that loans arranged by banks that tend to form larger syndicates are more likely to be granted on a secured basis. This is expected as there is room for shirking of monitoring duties by lead banks of larger syndicates which raises the demand for collateral from members. Expectedly, larger deals have more collateral requirements due to greater credit exposure of the syndicate, whereas loans issued by more reputed banks and to more profitable targets are less likely to have such conditions. Consistent with Sufi (2007), more concentrated syndicates require less collateral because of their more intense monitoring capacity. The coefficients of *Maturity* are positive and highly significant at the 1% level. This is consistent with Boot, Thakor and Udell (1991) who find that longer maturity loans are more likely to be secured given their higher default probabilities.

We address potential endogeneity concerns through tests similar to those reported in Panel A. The Durbin-Wu-Hausman χ^2 test statistics are all highly significant at the 1% level, suggesting that collateral is endogenous to LBO loan maturity. The Wald tests reject the weak instrument hypothesis and thereby confirm *Term Spread* as a valid instrument for LBO loan maturity.

³⁴ More details are available in Panel B of online Appendix E.

4.2.3. Spreads

Lastly, we test the effect of bank syndication networks on LBO loan pricing. We expect that a lead bank that is able to communicate better through its network should be able to bring down the total cost of borrowing.

We follow Bharath, Dahiya, Saunders and Srinivasan (2011) and posit that LBO loans are priced after the joint determination of loan maturity and collateral. The dependent variable *Spread* is the value-weighted interest rate spread over LIBOR that is charged on the LBO loan package.

Our specifications control for various deal-specific characteristics, prior ties between lead banks and borrowers, lead bank reputation and domicile, as these are known to significantly impact LBO loan terms (Demiroglu and James, 2010; Ivashina and Kovner, 2011). We further control for age of the PE firm and whether it is owned by other financial institutions. Next, we control for target firm quality using three accounting variables *Target Assets*, *Debt to Assets* and *EBITDA margin* based on balance sheet data obtained from Compustat.

Results from the analyses are reported in Panel C of Table 9. In model (2), the coefficients of *Diffusion* and *Sourcing* are both negative and significant at 1% level. In line with the previous results, traditional centralities are only marginally significant up to the 5% level. In fact, both *Degree* and *Eigenvector* drop out in the post-LASSO specification in model (5). This implies that information centralities have stronger effects on LBO loan pricing than their traditional counterparts. The ability of lead banks to diffuse and source information, and thereby reduce information asymmetries, allows the provision of cheaper LBO loans.

Panel C of online Appendix E in provides estimates of the economic significance of these coefficients. LBO borrowers pay a spread that is 13 (11) basis points lower for a one-standard-deviation increase in lead bank *Diffusion* (*Sourcing*). The mean cash savings from these lower interest rates, when discounted over the mean holding period (5.6 years) at a mean rate of 5.09% amounts to \$1.2 (\$1.02) million, respectively. This is equivalent to 0.6% (0.51%) higher return on investment for an equity contribution of \$198 million by the mean PE firm. In comparison, the other lead bank centralities have weaker effects on PE returns.

Consistent with literature, prior lending relationships are also associated with reduced spreads while target opacity (captured by *No Compustat Data*) leads to an increase in spreads. The exogenous instrument for loan spread, *Average 6m Spread*, behaves as expected with large, positive, and statistically significant effects at 1% across all models.

Diagnostics tests suggest that we cannot unconditionally reject the null of the exogeneity of maturity and collateral. The *p*-values of Durbin and Wu-Hausman tests are at most weakly significant at the 5% level. However, the first stage F-tests reject the null hypothesis of weak instruments across all the specifications.

Lastly, the coefficients of *Collateral* are negative but not statistically significant (through models (2)–(5)). As both non-price terms, *Maturity* and *Collateral*, are endogenous, we re-estimate the models after excluding these terms and obtained similar results. All our specifications include fixed effects for LBO deal year, target industry (two-digit SIC), and lead bank. Reported standard errors are heteroscedasticity robust and clustered at the bank–PE level.

5. Conclusions

Syndication among banks is pervasive throughout the LBO market, and to somewhat lesser extent in other corporate loan markets. Previous literature has emphasized the prevalence of relationship lending in these markets, and its central role in resolving information asymmetries between banks and borrowers. However, our analysis of loan-level data on US LBOs shows that lead banks had prior relationships with incumbent borrowers in at most 52% of the deals in our sample. What other channels might banks then utilize to resolve information problems during loan origination despite the scarceness of bank-borrower ties?

In this paper, we show that one such important and accessible channel is the network that develops as banks syndicate repeatedly with each other over time. Our findings show that the spread of information across the syndication network helps banks decide whether to join a syndicate, determine how much to contribute to the loan, and also to negotiate terms with prospective borrowers. We also show that the information flowing through the networks complements the information gained from previous interactions with either the LBO target or the PE sponsor. These results are economically significant: better information access via the network enables banks to issue cheaper LBO loans that provide major cost savings to borrowers.

Overall, our results highlight how network-based information flows might help resolve information problems. This concept was first developed for informal settings to understand how information transmits among members of village communities, or how information diffusion among voters facilitates political targeting. To our best knowledge, ours is the first study to show that these mechanisms play a similarly important role in formal settings such as loan syndication and contracting.

We identify several limitations and propose ideas for future research. Our approach does not allow us to observe actual network information flows. Analyzing the nature and quality of the actual information exchanged over the network, together with the nature and quality of prior interactions with borrowers, may provide better insights on how loan syndication actually occurs. Another interesting question is the extent to which banks are aware of the structure of the network they are embedded into. This is far from trivial given that banks are likely to make careful strategic choices in both how they acquire information and with whom they share it. This will further help relaxing the assumptions about the paths of information transmission and about the probability of this transmission. We leave these as open questions for future research.

References

- Abadie, A., and Imbens, G. W., 2002, Simple and bias-corrected matching estimators for average treatment effects, (National Bureau of Economic Research).
- Abadie, A., and Imbens, G. W., 2002, Simple and bias-corrected matching estimators for average treatment effects, (National Bureau of Economic Research).
- Altunbaş, Y., Gadanecz, B., and Kara, A., 2006. A Global Overview of the Syndicated Loans Market. (Palgrave Macmillan, London).
- Bae, K. H., and Goyal, V. K., 2009, Creditor rights, enforcement, and bank loans, *The Journal of Finance* 64, 823-860.
- Bain and Company, 2020, Global Private Equity Report 2020, (Bain & Company).
- Bajo, E., Chemmanur, T. J., Simonyan, K., and Tehranian, H., 2016, Underwriter networks, investor attention, and initial public offerings, *Journal of Financial Economics* 122, 376-408.
- Banerjee, A., Chandrasekhar, A. G., Duflo, E., and Jackson, M. O., 2013, The diffusion of microfinance, *Science* 341, 1236498.
- Banerjee, A., Chandrasekhar, A. G., Duflo, E., and Jackson, M. O., 2019, Using gossips to spread information: Theory and evidence from two randomized controlled trials, *The Review of Economic Studies* 86, 2453-2490.
- Belloni, A., and Chernozhukov, V., 2013, Least squares after model selection in high-dimensional sparse models, *Bernoulli* 19, 521-547.
- Belloni, A., Chernozhukov, V., and Hansen, C., 2014, High-dimensional methods and inference on structural and treatment effects, *Journal of Economic Perspectives* 28, 29-50.
- Belloni, A., Chernozhukov, V., and Hansen, C., 2014, Inference on treatment effects after selection among high-dimensional controls, *The Review of Economic Studies* 81, 608-650.
- Benmelech, E., Garmaise, M. J., and Moskowitz, T. J., 2005, Do liquidation values affect financial contracts? Evidence from commercial loan contracts and zoning regulation, *The Quarterly Journal of Economics* 120, 1121-1154.
- Berger, A., and Udell, G., 1995, Relationship lending and lines of credit in small firm finance, *Journal of Business* 68, 351-381.
- Berger, A. N., Espinosa-Vega, M. A., Frame, W. S., and Miller, N. H., 2011, Why do borrowers pledge collateral? New empirical evidence on the role of asymmetric information, *Journal of Financial Intermediation* 20, 55-70.
- Berger, A. N., and Udell, G. F., 1990, Collateral, loan quality and bank risk, *Journal of Monetary Economics* 25, 21-42.

- Bernstein, S., and Sheen, A., 2016, The operational consequences of private equity buyouts: Evidence from the restaurant industry, *The Review of Financial studies* 29, 2387-2418.
- Bharath, S. T., Dahiya, S., Saunders, A., and Srinivasan, A., 2011, Lending relationships and loan contract terms, *Review of Financial Studies* 24, 1141-1203.
- Bolton, P., Freixas, X., Gambacorta, L., and Mistrulli, P. E., 2016, Relationship and transaction lending in a crisis, *The Review of Financial Studies* 29, 2643-2676.
- Boot, A. W., Thakor, A. V., and Udell, G. F., 1991, Credible commitments, contract enforcement problems and banks: Intermediation as credibility assurance, *Journal of Banking & Finance* 15, 605-632.
- Boot, A. W. a., 2000, Relationship Banking: What Do We Know?, *Journal of Financial Intermediation* 9, 7-25.
- Brick, I. E., and Ravid, A. S., 1985, On the relevance of debt maturity structure., *The Journal of Finance* 40, 1423-1437.
- Bruche, M., Malherbe, F., and Meisenzahl, R., 2020, Pipeline Risk in Leveraged Loan Syndication, *The Review of Financial Studies*.
- Carey, M., and Nini, G., 2007, Is the corporate loan market globally integrated? A pricing puzzle, *The Journal of Finance* 2969-3007.
- Daniels, K., and Morgan, I., 2010, The Impact of Arrangers and Buyout Sponsors on Loan Pricing in LBOs: A Cross Border Study, in *Banking And Capital Markets: New International Perspectives* (World Scientific).
- Davis, S. J., Haltiwanger, J. C., Handley, K., Lipsius, B., Lerner, J., and Miranda, J., 2019, The Economic Effects of Private Equity Buyouts, (National Bureau of Economic Research).
- Degryse, H., and Ongena, S., 2005, Distance, lending relationships, and competition, *The Journal of Finance* 60, 231-266.
- Degryse, H., and Van Cayseele, P., 2000, Relationship lending within a bank-based system: Evidence from European small business data, *Journal of Financial Intermediation* 9, 90-109.
- Dehejia, R. H., and Wahba, S., 2002, Propensity score-matching methods for nonexperimental causal studies, *Review of Economics and statistics* 84, 151-161.
- Demiroglu, C., and James, C. M., 2010, The role of private equity group reputation in LBO financing, *Journal of Financial Economics* 96, 306-330.
- Dennis, S., Nandy, D., and Sharpe, L. G., 2000, The determinants of contract terms in bank revolving credit agreements, *Journal of Financial and Quantitative Analysis* 35, 87-110.

- Dennis, S. A., and Mullineaux, D. J., 2000, Syndicated loans, *Journal of Financial Intermediation* 9, 404-426.
- Dercon, S., 2005. Insurance against poverty (Oxford University Press).
- Diamond, D. W., 1984, Financial Intermediation and Delegated Monitoring, *The Review of Economic Studies* 51, 393-414.
- Diamond, D. W., 1989, Reputation acquisition in debt markets, *Journal of Political Economy* 97, 828-862.
- Diamond, D. W., 1991, Monitoring and reputation: The choice between bank loans and directly placed debt, *Journal of Political Economy* 689-721.
- Duarte, R., Finan, F., Larreguy, H., and Schechter, L., 2019, Brokering votes with information spread via social networks, (National Bureau of Economic Research).
- Ewens, M., and Farre-Mensa, J., 2019, The deregulation of the private equity markets and the decline in IPOs, (National Bureau of Economic Research).
- Fang, L., Ivashina, V., and Lerner, J., 2013, Combining banking with private equity investing, *Review* of Financial Studies 26.
- Farinha, L., and Santos, J., 2002, Switching from single to multiple bank lending relationships: Determinants and implications, *Journal of Financial Intermediation*.
- Fernando, C. S., May, A. D., and Megginson, W. L., 2012, The value of investment banking relationships: evidence from the collapse of Lehman Brothers, *The Journal of Finance* 67, 235-270.
- Fracassi, C., 2017, Corporate finance policies and social networks, *Management Science* 63, 2420-2438.
- Godlewski, C. J., and Sanditov, B., 2017, Financial Institutions Network and the Certification Value of Bank Loans, *Financial Management*.
- Godlewski, C. J., Sanditov, B., and Burger-Helmchen, T., 2012, Bank lending networks, experience, reputation, and borrowing costs: empirical evidence from the French syndicated lending market, *Journal of Business Finance & Accounting* 39, 113-140.
- Gompers, P., Kaplan, S. N., and Mukharlyamov, V., 2016, What do private equity firms say they do?, *Journal of Financial Economics* 121, 449-476.
- Graham, J. R., Li, S., and Qiu, J., 2008, Corporate misreporting and bank loan contracting, *Journal of Financial Economics* 89, 44-61.
- Hochberg, Y. V., Lindsey, L. A., and Westerfield, M. M., 2015, Resource accumulation through economic ties: Evidence from venture capital, *Journal of Financial Economics* 118, 245-267.

- Hochberg, Y. V., Ljungqvist, A., and Lu, Y., 2007, Whom you know matters: Venture capital networks and investment performance, *The Journal of Finance* 62, 251-301.
- Hochberg, Y. V., Ljungqvist, A., and Lu, Y., 2010, Networking as a barrier to entry and the competitive supply of venture capital, *The Journal of Finance* 65, 829-859.
- Ivashina, V., 2009, Asymmetric information effects on loan spreads, *Journal of Financial Economics* 92, 300-319.
- Ivashina, V., and Kovner, A., 2011, The private equity advantage: Leveraged buyout firms and relationship banking, *Review of Financial Studies* 24, 2462-2498.
- Ivashina, V., and Scharfstein, D., 2010, Loan syndication and credit cycles, American Economic Review 100, 57-61.
- Kaplan, S. N., and Strömberg, P., 2009, Leveraged buyouts and private equity, *The Journal of Economic Perspectives* 23, 121-146.
- Kobayashi, T., and Takaguchi, T., 2018, Identifying relationship lending in the interbank market: A network approach, *Journal of Banking & Finance* 97, 20-36.
- Lerner, J., 1994, The syndication of venture capital investments, Financial Management 16-27.
- Li, D., and Schürhoff, N., 2019, Dealer networks, The Journal of Finance 74, 91-144.
- Ljungqvist, A., Richardson, M., and Wolfenzon, D., 2020, The investment behavior of buyout funds: Theory and evidence, *Financial Management*.
- López-Espinosa, G., Mayordomo, S., and Moreno, A., 2017, When does relationship lending start to pay?, *Journal of Financial Intermediation* 31, 16-29.
- McFadden, D. L., 1984, Econometric analysis of qualitative response models, *Handbook of Econometrics*, Vol. 2, 1395-1457.
- Melnik, A., and Plaut, S., 1986, Loan commitment contracts, terms of lending, and credit allocation, *The Journal of Finance* 41, 425-435.
- Ozmel, U., Robinson, D. T., and Stuart, T. E., 2013, Strategic alliances, venture capital, and exit decisions in early stage high-tech firms, *Journal of Financial Economics* 107, 655-670.
- Papke, L. E., and Wooldridge, J. M., 1996, Econometric methods for fractional response variables with an application to 401 (k) plan participation rates, *Journal of Applied Econometrics* 11, 619-632.
- Petersen, M. A., and Rajan, R. G., 1994, The benefits of lending relationships: Evidence from small business data, *The Journal of Finance* 49, 3-37.

- Petersen, M. A., and Rajan, R. G., 1995, The effect of credit market competition on lending relationships, *The Quarterly Journal of Economics* 110, 407-443.
- Plagmann, C., and Lutz, E., 2019, Beggars or choosers? Lead venture capitalists and the impact of reputation on syndicate partner selection in international settings, *Journal of Banking & Finance* 100, 359-378.
- Qian, J., and Strahan, P. E., 2007, How laws and institutions shape financial contracts: The case of bank loans, *The Journal of Finance* 62, 2803-2834.
- Rajan, R. G., 1992, Insiders and outsiders: The choice between informed and arm's-length debt, *The Journal of Finance* 47, 1367-1400.
- Richmond, R. J., 2019, Trade network centrality and currency risk premia, *The Journal of Finance* 74, 1315-1361.
- Robinson, D. T., and Stuart, T. E., 2007, Network effects in the governance of strategic alliances, Journal of Law, Economics, and Organization 23, 242-273.
- Rosenbaum, P. R., and Rubin, D. B., 1983, The central role of the propensity score in observational studies for causal effects, *Biometrika* 70, 41-55.
- Ross, D. G., 2010, The "dominant bank effect:" How high lender reputation affects the information content and terms of bank loans, *Review of Financial Studies* 23, 2730-2756.
- Saunders, A., and Steffen, S., 2011, The costs of being private: Evidence from the loan market, *Review of Financial Studies* 24, 4091-4122.
- Saunders, A., Steffen, S., Freudenberg, F., and Imbierowicz, B., 2012, Covenant violations, loan contracting, and default risk of bank borrowers.
- Sharpe, S. A., 1990, Asymmetric Information, Bank Lending, and Implicit Contracts: A Stylized Model of Customer Relationships, *The Journal of Finance* 45, 1069-1087.
- Sorenson, O., and Stuart, T. E., 2001, Syndication networks and the spatial distribution of venture capital investments, *American Journal of Sociology* 106, 1546-1588.
- Staiger, D., and Stock, J. H., 1997, Instrumental Variables Regression with Weak Instruments, *Econometrica* 65, 557-586.
- Sufi, A., 2007, Information asymmetry and financing arrangements: Evidence from syndicated loans, *The Journal of Finance* 62, 629-668.
- Udry, C. R., and Conley, T. G., 2004, Social networks in Ghana, *The Social Economics of Poverty* 232.

Figures and Tables

Figure 1. LBO loan syndication process

The loan syndication process for a leveraged buyout (LBO) typically begins after a lead bank (or a group of lead banks in some cases) is chosen to arrange a loan for the deal. Subsequently, the lead bank solicits other banks in its network to sell parts of the loan to them. During this process, the lead bank collects information on the borrowers (i.e. LBO target and PE firm) and conducts due diligence. Based on its assessment, the lead bank submits a confidential memorandum to prospective syndicate members (participants), containing details of borrowers and an assessment of the deal. Interested banks will negotiate the terms of their participation and sign letters of commitment with the lead bank. Upon forming the syndicate, the lead bank negotiates the terms of the LBO loan package (such as loan maturity, collateral requirements, and interest rate) with the borrowers. Post syndication, the lead bank is responsible for monitoring the borrowers and communicating their performance to the members. Thus, the lead bank acts as an agent on behalf of the lending syndicate. For a detailed description of the loan syndication process, see Altunbas *et al.* (2006), Sufi (2007), Ivashina (2009), and, Bharath *et al.* (2011).



Figure 2: Active banks by year in the US LBO loan market

Figure shows the number of banks active in the US LBO loan market by year. A bank is considered active in year *t* if it was involved in at least one LBO in the preceding five years, i.e. between t-5 and *t*. Blue line represents banks that are domiciled or headquartered in the US. Red line represents banks domiciled or headquartered outside the US. Black line denotes all banks irrespective of domicile or headquarter location.



Figure 3: Bank syndication networks in the US LBO market (1990–2012)

Figure presents the ties between banks active in the US LBO loan market during a given year (as shown in the title of each plot). Two banks *i* and *j* are considered to have a tie in year *t* if they were part of at least one LBO loan syndicate in the preceding five years, i.e. between t-5 and *t*. The network layout is constructed using the Fruchterman-Reingold algorithm. Blue dots represent banks that are domiciled or headquartered in the US. Red dots represent banks domiciled or headquartered outside the US.



Figure 4: Mean network centralities by year

Figure presents the distributional properties of bank syndication networks by year in the US LBO loan market. The mean network centralities shown below are computed over a rolling five year window. For each year *t*, *centrality_{it}* of bank *i* is computed based on its syndication activity with other banks in the period t-5 and t.



Figure 5: Skewness of network centralities by year

Figure presents the distributional properties of bank syndication networks by year in the US LBO loan market. The skewness of network centralities shown below are computed over a rolling five year window. For each year *t*, *centrality*_{it} of bank *i* is computed based on its syndication activity with other banks in the period t-5 and *t*. Skewness of a variable represents the degree of asymmetry in its probability distribution relative to the normal distribution.



Figure 6. Do bank networks complement or substitute for borrower relationships?

Plots present marginal effects of the interaction terms reported in Table 7. The y-axis in each plot denotes the probability of a candidate bank joining an LBO loan syndicate as non-lead member. Each plot shows the interaction effect between the network centrality of a candidate bank (*Cand Centrality*) and a discretized version of prior borrowing relationships between that bank and LBO borrowers (i.e. LBO target or PE firm). Plots (a) and (b) are based on coefficients of interaction terms in model (1). Plots (c) and (d) are based on coefficients of interaction terms in model (3). Plots (g) and (h) are based on coefficients of interaction terms in model (4). Vertical bars represent the 95% confidence intervals at each level of the running variable (x-axis).



37

Variable	Definition
LBO loan characteristics	
Match (D)	Dummy equal to one if a candidate bank joins as a non-lead member in an LBO loan syndicate.
Spread (%)	Loan spread in percentage points, weighted by the value of each loan tranche relative to the total deal size. Spreads represent the total interest rate (including fees) paid in excess of LIBOR on the loan package. We use a value-weighted measure of spread since LBO deals typically comprise several loan tranches in an increasing order of seniority, with each tranche having different borrowing terms and characteristics.
Maturity	Loan maturity in months, weighted by the value of each loan tranche relative to the total deal size.
Collateral	Dummy term equal one if the LBO loan has any collateral requirements.
Max Debt to EBITDA	Covenant imposed by the bank syndicate on the LBO loan stating the maximum debt to EBITDA ratio that the target must maintain throughout the duration of the loan.
Min Interest Coverage	Covenant imposed by the bank syndicate on the LBO loan stating the minimum interest coverage ratio that the target must maintain throughout the duration of the loan.
Deal Size	Total size of the LBO loan (in mil. USD).
Term Spread	This variable is used an instrument for LBO loan maturity. Term spread is the difference in yields on ten-year and one-year US government bonds at the time of LBO loan issue as reported by the US Federal Reserve.
LBO loan syndicate characteristics	
Syndicate Size	Number of banks participating in the LBO loan syndicate (including lead bank).
Members	Number of non-lead member banks participating in the LBO loan syndicate.
Lead Bank Share (%)	Total share of the lead bank(s) in the LBO loan.
Debt/EBITDA	Total LBO debt divided by pre-deal EBITDA of the target.
Syndicate Herfindahl	Sum of squared percentage share of each bank in the LBO loan syndicate.
Bank network centralities (All centre	alities are measured over the five years preceding the LBO)
Diffusion	Expected number of times that information originated by bank <i>i</i> is heard by other banks in the network over the time period <i>T</i> . Suppose information pertaining to an LBO originates from participating bank <i>i</i> , and is broadcast in the first period $t = 1$ among <i>i</i> 's neighbors with probability <i>p</i> . During each subsequent period, every informed neighbor shares this information and the identity of the source with its neighbors with the same probability <i>p</i> . This process is repeated over <i>T</i> time periods. The hearing matrix H is then defined as:
	$\mathbf{H}(\boldsymbol{g}_{y}, \boldsymbol{p}, T) = \sum_{t=1}^{t} (\boldsymbol{p}\boldsymbol{g}_{y})^{t}$
	where g is the <i>N</i> × <i>N</i> undirected adjacency matrix of the bank network for a given year <i>y</i> . The <i>ij</i> - th element of H is the expected number of times that bank <i>j</i> hears some information that originated from bank <i>i</i> over <i>T</i> previous periods. Banerjee <i>et al.</i> (2013) show that the per-period transmission probability, <i>p</i> , can be reasonably well approximated by $1/E[\lambda_1(g)]$, which is the inverse of the largest eigenvalue of the adjacency matrix <i>g</i> . They also suggest that <i>T</i> can be approximated by the diameter of the network. <i>Diffusion</i> centrality is then defined as:
	$Diffusion(\boldsymbol{g}_{y}, \mathbf{p}, \mathbf{T}) = \mathbf{H}(\boldsymbol{g}_{y}, \mathbf{p}, \mathbf{T}) \cdot 1 = \sum_{x}^{T} (p\boldsymbol{g}_{y})^{t} \cdot 1$
Sourcing	Expected number of times that bank <i>j</i> "hears" information from the other banks over <i>T</i> periods. Suppose in each period, a bank receives information originating from various parts of the network. Since $\mathbf{H}(\boldsymbol{g}_{y}, p, T)_{ij}$ is the expected number of times bank <i>j</i> hears information originating from bank <i>i</i> , the <i>j</i> -th column of H denotes bank <i>j</i> 's expescted information sourcing from every other bank in the network. Sourcing centrality of bank <i>j</i> is thus the average of the <i>j</i> -th column of H .
Degree	Number of unique connections of the bank with other banks in the LBO market.
Eigenvector	Number of unique connections of the bank with other banks in the LBO market, where each connection is weighted by the eigenvector centrality of the other bank. Represents the bank's ties to prominent and well-connected banks within the network.
Network Distance	Length of the shortest path between two banks across the LBO loan syndication network.

Table 1. Variable definitions

	(Continues next page)
PE characteristics	
PE Captive (D)	Dummy equal to one if the PE firm is significantly owned and controlled by a financial institution (typically banks or insurance companies), and zero otherwise.
PE Age	Age of the PE-sponsor in years.
PE Reputation	Total amount of equity funds raised by the PE firms in the five years preceding an LBO deal.
Bank characteristics (All historical w	variables are estimated over the five-year period preceding the LBO)
Bank-PE Relationship	Total volume of LBO loans issued to the PE firm over the last five years in which the incumbent bank participated either as a lead bank or member. This measure is similar to the variable <i>Bank relationship (amount)</i> used by Ivashina & Kovner (2011), and applies to the <i>Lead-PE Relationship</i> and <i>Member-PE Relationship</i> variables used in the empirical analysis.
Bank-Target Relationship	Total volume of non-LBO loans issued to the target over the last five years in which the incumbent bank participated either as a lead bank or member. This variable excludes all debt issued as part of the focal LBO transaction, and applies to the <i>Lead-Target Relationship</i> and <i>Member-Target Relationship</i> variables used in the empirical analysis.
Bank-Bank Relationship	Total volume of past LBO loans in the past five years in which two banks <i>i</i> and <i>j</i> were part of the same loan syndicate. This measure applies to the <i>Lead-Member Relationship</i> variable used in the empirical analysis.
Non US Bank (D)	Dummy equal to one if the bank is domiciled or headquartered outside the US.
Bank Lead Experience (%)	Percentage of participated LBO loans in the last five years in which the bank was the lead arranger.
Bank LBO Market Share (%)	Percentage of all LBO loans (in mil. USD) during the previous five years in which the bank participated either as lead or member.
Bank-PE Geodist	Great circle geographic distance between the headquarters of the bank and PE firm in kilometers.
Bank-Bank Geodist	Great circle geographic distance between the headquarters of two given banks in kilometers. This measure applies to the <i>Bank-Lead Geodist</i> variable used in the empirical analysis.
Avg Number of Members	Average number of banks that participated as members in previous LBO loans arranged by the incumbent bank.
Average 6m Spread	Lagged average spread charged by banks on all LBO loans issued during the previous six months, based on Bharath et <i>al.</i> (2011).
Target characteristics	
Age	Age of the LBO target in years
Assets	Book value of total assets of the target in the year of the LBO.
LT Debt to assets (%)	Ratio of long-term debt to total assets of the target in the year of the LBO.
EBITDA (%)	Ratio of EBITDA to sales of the target prior in the year the LBO.
No Compustat Data (D)	Dummy equal to one if Compustat data is not available for target.

Table 2. Summary statistics for bank characteristics

Table presents summary statistics of network centralities of banks active in the US leveraged buyout (LBO) loan market during the period 1991-2012. A US LBO loan refers to syndicated loans sponsored by US-based PE firms to acquire US-based targets. Syndicate members refers to non-lead (*member*) banks active in this market during this period. A bank is included in the sample if it was active in the US LBO loan market during the five-year trailing period. Data on US LBO syndicated loans comes from LPC Dealscan database and ThomsonOne. See Table 1 for variables description.

	L	Lead Banks			ate Memb	ers	T-Test
	Ν	Mean	SD	Ν	Mean	SD	(difference of means)
Diffusion	2746	8.05	7.35	5799	3.23	3.53	23.516***
Sourcing	2746	0.08	0.05	5796	0.10	0.06	-6.834***
Degree	2746	1.22	0.87	5796	0.83	0.57	13.971***
Eigenvector	2746	0.44	0.29	5794	0.28	0.18	15.072***

Table 3. Pairwise correlations between centrality measures

Table presents pairwise correlations between network centralities of banks that syndicated in the US leveraged buyout (LBO) loan market during the period 1991-2012. A US LBO loan refers to syndicated loans sponsored by US-based PE firms to acquire US-based targets. See Table 1 for variables description.

-		Diffu	So	De
		sion	urcing	gree
S	ourc			
ing		0.044***		
D	egre		0.3	
e		0.258***	30***	
E	igen		0.4	0.8
vector		0.322***	84***	94***

Table 4. Summary statistics for LBO characteristics

Table presents summary statistics for a sample of US leveraged buyout (LBO) loans during the period 1991-2012. A US LBO loan refers to syndicated loans sponsored by US-based PE firms to acquire US-based targets. The primary data sources on LBO deal terms, and bank and PE firm characteristics are the LPC Dealscan database and ThomsonOne. Data on target firm characteristics prior to the LBO is from Compustat. See Table 1 for variables description.

Variable	Ν	Mean	SD	Min	Median	Max
LBO Deal Characteristics						
Deal Value (\$m)	2807	303.5	395.1	24	207	5525
Loan Spread over LIBOR (%)	2834	3.8	1.7	0	3.7	11.5
Loan Maturity (months)	2646	66.8	14.8	12	68.0	120
Loan Collateral (D)	2859	0.5	0.5	0	0	1
Max Debt to EBITDA	424	5.8	1.5	1.75	5.9	11.3
Min Interest Coverage	346	1.9	0.5	1	1.8	5.0
LBO Debt/EBITDA	491	3.6	4.1	0.1	2.6	34.3
Syndicate Size	2834	4.2	2.4	1	4	14
No Members	2832	2.3	2.3	0	2	12
Lead Bank Share (%)	2237	41.9	22.8	0	40	98
Non-US Bank (D)	2859	0.7	0.4	0	1	1
Bank Characteristics						
Lead-PE Relationship (\$m)	2833	108.1	350.9	0	0	5005.0
Lead-Target Relationship (\$m)	2831	14.3	78.3	0	0	2288.3
Member-PE Relationship (\$m)	1744	79.8	252.7	0	0	1662.2
Member-Target Relationship (\$m)	1744	6.7	22.6	0	0	144.2
Lead-Member Relationship (\$m)	1744	2048.3	2935.9	0	710	13498.1
Bank Lead Experience (%)	2731	49.4	25.3	0	50.9	89.2
Bank LBO Market Share (%)	2703	1.0	1.4	0.0	0.6	7.9
PE Characteristics						
PE Captive (D)	2859	0.08	0.27	0	0	1
PE Age (years)	2245	18.7	15.5	2	17	48
Target Characteristics						
Target Age (years)	2379	36.5	22.8	1	37	98
Assets (\$m)	206	1478.8	2679.9	22.4	810.3	19115.6
EBITDA (%)	206	16	10	-5	13	59
Debt to Assets (%)	206	42	28	0	42	118
No Compustat Data (D)	2377	0.91	0.25	0	1	1

Table 5. Relationship between network centralities and bank reputation

Table presents OLS estimates of the relationship between bank network centralities and market share in the US LBO loan market using the following specification:

$$Centrality_{it} = \beta_0 + \beta_1 Bank \ LBO \ Market \ Share_{it} + \alpha_i + \mu_t + \varepsilon_{it}$$

*Centrality*_{*it*} denotes any of our four centrality measures of bank *i* in year *t*, *Bank LBO Market Share*_{*it*} is the proportion of LBO loan syndicates (in dollar terms) in which bank *i* participated during the five years preceding year *t*. a_i and μ_t is a vector of bank and year fixed effects respectively, and ε_{it} is the error term. The sample spans the period 1991-2012 and is based on adjacency matrices that represent dyadic ties between any two banks in the loan syndication network. Since this network evolves continuously due to the changes in the syndication process, market trends, and the entry/exit of banks, we construct adjacency matrices on an annual rolling basis using five-year trailing windows. These adjacency matrices are then used to compute the relevant network centralities for each bank in the network. Banks that were not active in the market over the five-year trailing period are considered newcomers and have their network centralities set to zero. See Table 1 for variables description.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Diffusion	Sourcing	Degree	Eigenvector	Diffusion	n Sourcing	Degree	Eigenvector
Bank LBO Mkt Share	-0.098	0.232	0.211**	0.272***	-0.096	0.231	0.214**	0.274***
	(0.060)	(0.174)	(0.092)	(0.052)	(0.059)	(0.172)	(0.093)	(0.052)
Bank FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time FE	No	No	No	No	Yes	Yes	Yes	Yes
Adjusted R ²	0.424	0.131	0.209	0.211	0.433	0.209	0.232	0.248
Observations	22490	22127	22472	22490	22490	22127	22472	22490

Table 6. Bank network centralities and LBO loan syndicate participation

Table reports conditional logistic (CL) estimates of the probability of a candidate bank's participation as non-lead member of an LBO loan syndicate according to the following specification:

$$Pr(Match_{ij} = 1) = \beta_0 + \beta_1 \cdot Cand \ Centrality_i + Controls'_{ij} \cdot \beta_2 + \mu_j + \alpha_t + \varepsilon_{ij}$$

The dependent variable *Match_{ij}* is a dummy term equal to one if bank *i* was a member of the LBO syndicate for deal *j*, and zero otherwise. *Cand Centrality_i* denotes any of the four centrality measures for a candidate bank *i* depending on the specification. *Controls_{ij}* is a set of control variables capturing other observable characteristics such as prior relationships between bank *i* and the incumbent borrowers that could influence syndicate participation choice. μ_j and α_t denote deal and time fixed effects while ε_{ij} is the error term. The sample is constructed using case-control matching, where for each deal, we identify up to 10 candidate banks that are closest to the incumbent members in terms of their propensity scores based on observable characteristics. The matching is done with replacement (so that candidates can be used for matching more than once) as it reduces bias in the estimates. The sample period is from 1991-2012 and is at the deal level. See Table 1 for variables description. Coefficients in columns (1)-(4) are reported as odds ratios. Numbers in parentheses are robust standard errors adjusted for heteroscedasticity and clustering at the deal level. ***, **, and * denote that the coefficient is significantly different from zero at 1%, 5%, and 10% level (two-tailed), respectively.

	(1)	(2)	(3)	(4)
	CI	CI	CI	Post-LASSO
	CL	CL	CL	OLS
Diffusion	1.415***			0.008
	(0.034)			(0.005)
Sourcing	1.839***			0.053***
	(0.081)			(0.006)
Degree		2.184***		
		(0.076)		
Eigenvector			2.149***	0.071***
-			(0.070)	(0.006)
Ln(Bank–Lead Relationship)	1.521***	1.374***	1.359***	0.006*
	(0.049)	(0.045)	(0.044)	(0.003)
Ln(Bank–PE Relationship)	1.449***	1.377***	1.372***	0.042***
	(0.038)	(0.037)	(0.037)	(0.004)
Ln(Bank–Target Relationship)	1.406***	1.387***	1.391***	0.058***
	(0.034)	(0.035)	(0.035)	(0.003)
Non US Bank	0.987	0.873**	0.858***	
	(0.058)	(0.051)	(0.049)	
Ln(Bank-PE Geodist)	0.971***	0.970***	0.971***	-0.004***
	(0.010)	(0.011)	(0.011)	(0.001)
Ln(Bank-Lead Geodist)	0.993	0.991	0.994	-0.003*
	(0.012)	(0.012)	(0.012)	(0.001)
Constant				0.277***
				(0.036)
Time FE				Yes
Target Industry FE				Yes
Pseudo/Adjusted R ²	0.178	0.178	0.179	0.107
χ^2	1880.787	1756.640	1849.924	37.95
p-value	0.000	0.000	0.000	0.000
Observations	21,112	21,112	21,112	21,112

Table 7. Bank network centralities and LBO loan syndicate participation: impact of relationships

Table reports conditional logistic (CL) estimates of the probability of a candidate bank's participation as non-lead member of an LBO loan syndicate according to the following specification:

 $\begin{aligned} & Pr(Match_{ij} = 1) = \beta_0 + \beta_1 \cdot Cand \ Centrality_i + \beta_2 \cdot Cand \ Relationship_i + \\ & \beta_3 \cdot Cand \ Centrality_i * Cand \ Relationship_i + Controls'_{ij} \cdot \beta_4 + \mu_j + \alpha_t + \varepsilon_{ij} \end{aligned}$

The dependent variable *Match_{ij}* is a dummy term equal to one if bank *i* was a member of the LBO syndicate for deal *j*, and zero otherwise. *Cand Centrality_i* denotes any of the four centrality measures for a candidate bank *i* depending on the specification. *Cand Relationship_i* denotes prior borrowing relationships between candidate bank *i* and LBO borrowers (i.e. LBO target or PE firm). *Controls_{ij}* is a set of control variables capturing other observable characteristics such as prior relationships between bank *i* and the incumbent borrowers that could influence syndicate participation choice. μ_j and α_t denote deal and time fixed effects while ε_{ij} is the error term. The sample is constructed using case-control matching, where for each deal, we identify up to 10 candidate banks that are closest to the incumbent members in terms of their propensity scores based on observable characteristics. The matching is done with replacement (so that candidates can be used for matching more than once) as it reduces bias in the estimates The sample period is from 1991-2012 and is at the deal level. See Table 1 for variables description. Coefficients in columns (1)-(4) are reported as odds ratios. Numbers in parentheses are robust standard errors adjusted for heteroscedasticity and clustering at the deal level. ***, **, and * denote that the coefficient is significantly different from zero at 1%, 5%, and 10% level (two-tailed), respectively.

	(1)	(2)	(3)	(4)	(5)
	CL	CL	CL	CL	Post-LASSO
	ee.	CE	CE	CE	OLS
Diffusion	1.504***	1.420***			
~ .	(0.040)	(0.033)			
Sourcing	1.507***	1.493***			0.024***
	(0.059)	(0.058)			(0.006)
Degree			2.413***		
			(0.100)		
Eigenvector				2.242***	0.061***
				(0.084)	(0.006)
Ln(Bank–Lead Relationship)	1.585***	1.629***	1.340***	1.328***	
	(0.051)	(0.051)	(0.045)	(0.044)	
Ln(Bank–PE Relationship)	1.528***	1.441***	1.460***	1.452***	0.051***
	(0.040)	(0.037)	(0.041)	(0.041)	(0.004)
Ln(Bank–Target Relationship)	1.455***	1.409***	1.458***	1.450***	0.074***
	(0.034)	(0.034)	(0.035)	(0.035)	(0.003)
Diffusion × Ln(Bank–PE Relationship)	0.937***				
	(0.011)				
Diffusion × Ln(Bank–Target Relationship)	0.946***				
	(0.011)				
Sourcing × Ln(Bank–PE Relationship)		1.108^{***}			0.022***
		(0.029)			(0.004)
Sourcing × Ln(Bank–Target Relationship)		0.984			0.004
		(0.023)			(0.004)
Degree × Ln(Bank–PE Relationship)			0.866***		
			(0.020)		
Degree × Ln(Bank–Target Relationship)			0.885***		
			(0.018)		
Eigenvector × Ln(Bank–PE Relationship)				0.889^{***}	
				(0.019)	
Eigenvector × Ln(Bank–Target Relationship)				0.904***	-0.015***
				(0.017)	(0.003)
Constant					0.134
					(0.102)
Controls	Yes	Yes	Yes	Yes	Yes
Pseudo/Adjusted R ²	0.179	0.177	0.161	0.157	0.108
χ^2	2171.028	1964.079	1959.324	1910.709	32.98
p-value	0.000	0.000	0.000	0.000	0.000
Observations	21,112	21,112	21,112	21,112	21,112

Table 8. Network centralities and lead bank share (%)

Table reports OLS estimates of the relationship between lead bank network centralities and their percentage shareto the LBO loan using the following specification:

Lead Bank Share_j = $\beta_0 + \beta_1 \cdot Lead Centrality_j + \beta_2 \cdot Opaque + \beta_3 \cdot Lead Centrality_j * Opaque + Controls'_j \cdot \beta_4 + \gamma_i + \alpha_t + \varepsilon_j$

The dependent variable *Lead Share_j* is the percentage of the LBO loan retained by the lead bank in deal *j*. *Lead Centrality* denotes any of our four centrality measures for the lead bank. *Opaque* represents the extent to which the bank syndicate must assess and monitor the borrower, and is proxied by *PE Reputation*, measured as the total amount of funds raised by the PE firm in the five years preceding the LBO. *Controls_j* is a set of control variables for observable deal and borrower characteristics. γ_{ij} and α_t denote target industry and time fixed effects while ε_{it} is the general error term. The sample period is from 1991-2012 and is at the deal level. See Table 1 for variables description. Numbers in parentheses are robust standard errors adjusted for heteroscedasticity and clustering at the deal level. ***, **, and * denote that the coefficient is significantly different from zero at 1%, 5%, and 10% level (two-tailed), respectively.

	(1)	(2)	(3)	(4)
Diffusion	-0.082***	-0.086***	-0.082***	-0.089***
	(0.014)	(0.014)	(0.014)	(0.014)
Sourcing	-0.066***	-0.066***	-0.082***	-0.084***
	(0.013)	(0.013)	(0.015)	(0.015)
PE Reputation	0.035*	0.037*	0.020	0.020
	(0.019)	(0.021)	(0.018)	(0.019)
Diffusion × PE Reputation		-0.040		-0.060**
-		(0.034)		(0.029)
Sourcing × PE Reputation			-0.081***	-0.095***
- •			(0.026)	(0.029)
Ln(Lead–PE Relationship)	0.082***	0.082***	0.081***	0.080***
-	(0.019)	(0.019)	(0.019)	(0.019)
Ln(Lead-Target Relationship)	0.058***	0.058***	0.061***	0.061***
	(0.016)	(0.016)	(0.016)	(0.016)
Ln(Lead-Member Relationship)	-0.004	-0.004	-0.003	-0.003
• *	(0.015)	(0.015)	(0.015)	(0.015)
Ln(Member–PE Relationship)	-0.117***	-0.117***	-0.117***	-0.116***
• *	(0.019)	(0.019)	(0.019)	(0.019)
Ln(Member-Target Relationship)	-0.036**	-0.036**	-0.038**	-0.039**
	(0.017)	(0.017)	(0.017)	(0.017)
Ln (Deal Val)	0.033**	0.034**	0.033**	0.033**
	(0.015)	(0.015)	(0.015)	(0.015)
Lead Non-US	-0.003	-0.002	-0.000	0.001
	(0.034)	(0.034)	(0.034)	(0.034)
Ln (Target Assets)	-0.008	-0.006	-0.012	-0.009
	(0.061)	(0.060)	(0.060)	(0.060)
Ln (Target EBITDA)	-0.129	-0.125	-0.105	-0.094
	(0.513)	(0.519)	(0.504)	(0.510)
Target LT Debt to Assets	-0.151	-0.152	-0.154	-0.156
	(0.208)	(0.208)	(0.208)	(0.208)
No Compustat Data	-0.020	-0.026	-0.016	-0.024
	(0.072)	(0.072)	(0.072)	(0.071)
Constant	-0.124	-0.138	-0.111	-0.129
	(0.460)	(0.457)	(0.458)	(0.453)
Target Industry FE	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes
Observations	1603	1603	1603	1603

Table 9. Lead bank network centralities and LBO borrowing terms

Table reports instrumental variables estimates of the following structural model that investigate the effects of the network centrality of lead banks on LBO loan terms:

 $\begin{aligned} Maturity &= \beta_{10} + \beta_{11} Centrality + \beta_{12} Collateral + Controls'_k \beta_{13} + X'_1 \beta_{14} + \varepsilon_1 \\ Collateral &= \beta_{20} + \beta_{21} Centrality + \beta_{22} Maturity + Controls'_k \beta_{23} + X'_2 \beta_{24} + \varepsilon_2 \end{aligned}$

 $Spread = \beta_{30} + \beta_{31}Centrality + \beta_{32}Maturity + \beta_{33}Collateral + Controls'_k\beta_{34} + X'_3\beta_{35} + \varepsilon_3$

The dependent variable in Panel A is the natural logarithm of LBO loan maturity in months. The dependent variable in Panel B is collateral, which equals 1 if the deal was secured with collateral, and 0 otherwise. The dependent variable in Panel C is LBO loan spread measured in percentage points. Spreads represent the total interest rate (including fees) paid in excess of LIBOR on the loan package. Since LBO deals are typically structured using several loan tranches, all the three dependent variables are valueweighted measures, weighted by the value of each loan tranche relative to the total deal size. We use the instrumental variables approach to analyze the above system of equations and introduce X_i as exogenous instruments to control for potential endogeneity of the corresponding dependent variable. Term spread is the difference in yields on one-year and ten-year US government bonds prevailing at the time of LBO deal origination as reported by the US Federal Reserve. Syndicate Herfindahl represents moral hazard within the LBO debt syndicate and is measured using the Herfindahl-Hirschman methodology as the sum of squared percentage shareof each participating bank (including lead banks) to the LBO debt package. The Durbin-Wu tests in Panels A and C determine whether loan maturity and collateral are endogenous in the corresponding models. A rejection of the null hypothesis implies that they must be treated as endogenous variables. The Wald's chi-square test provides a similar test of endogeneity for the IV probit models in Panel B. To test for instrument validity in Panels A and C, we report the first-stage F-statistic that determines the joint significance of the included instruments. A significant F-statistic (p < 0.00) implies that our instruments have considerable explanatory power for the endogenous variables. Average 6m spread is the lagged average spread charged by banks on all LBO loans issued during the previous six months, following Bharath et al. (2011). See Table 1 for variables description. Numbers in parentheses are robust standard errors adjusted for heteroscedasticity and clustering at the bank-PE level. ***, **, and * denote that the coefficient is significantly different from zero at 1%, 5%, and 10% level (two-tailed), respectively.

Panels A, B, and C are on separate pages.

	(1)	(2)	(3)	(4)	(5)
	IV	IV	IV	IV	Post-LASSO OLS
Diffusion		0.017***			0.023***
		(0.006)			(0.006)
Sourcing		0.005			-0.000
		(0.004)			(0.005)
Degree			-0.001		
			(0.004)		
Eigenvector				0.003	
				(0.004)	
Ln(Lead-PE Relationship)	0.003	0.001	0.003	0.002	
	(0.002)	(0.002)	(0.003)	(0.003)	
Ln(Lead-Target Relationship)	-0.001	-0.002	-0.001	-0.001	
	(0.004)	(0.004)	(0.005)	(0.005)	0.005
Avg Number of Members	0.016	0.016	0.016	0.016	0.027***
	(0.011)	(0.011)	(0.010)	(0.011)	(0.008)
Non US Lead	-0.021	-0.024	-0.021	-0.023	
	(0.019)	(0.018)	(0.018)	(0.018)	
Ln (Deal Val)	-0.01/	-0.015	-0.01/	-0.016	
	(0.014)	(0.014)	(0.014)	(0.014)	
Captive PE	0.001	-0.001	0.001	0.000	
	(0.022)	(0.022)	(0.022)	(0.022)	
Ln (PE Age)	-0.010	-0.009	-0.011	-0.010	
Land LDO Mitt Shaw	(0.008)	(0.008)	(0.007)	(0.007)	
Lead LBO Wikt Share	-0.000	-0.000	-0.000	-0.000	
	(0.000)	(0.000)	(0.000)	(0.000)	
Ln (Target Assets)	-0.031	-0.030*	-0.031	-0.031	
L_{n} (Torrat EDITDA)	(0.021)	(0.021)	(0.021)	(0.021)	
LII (Target EBITDA)	(0.100)	(0.192)	(0.100)	(0.100)	
Torract I T Dabt to Accesta	(0.190)	(0.185)	(0.190)	(0.190)	0.170***
Target LT Debt to Assets	(0.131)	(0.072)	(0.072)	(0.126)	(0.057)
No Compustat Data	(0.073)	(0.073)	(0.073)	(0.073)	(0.057)
No Compustat Data	(0.010)	(0.020)	(0.017)	(0.010)	
Loon Collatoral	(0.032)	(0.031)	(0.052)	0.0052)	0.063
Loan Conaterai	(0.120)	(0.126)	(0.119)	(0.118)	(0.072)
Term Spread	0.055***	0.051***	0.055***	0.055***	0.055***
Term Spread	(0.007)	(0.006)	(0.007)	(0.007)	(0.005)
Constant	4 251***	4 305***	4 251***	4 252***	4 224***
Constant	(0.175)	(0.181)	(0.176)	(0.174)	(0.079)
Target Industry FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
Adjusted R^2	0.226	0.247	0.226	0.227	0.203
Durbin γ^2 -test	0.778	0.423	0.802	0.753	1.100
p-value (Durbin γ^2 -test)	0.378	0.515	0.370	0.386	0 294
Wu-Hausman F-test	0 748	0.406	0.771	0.723	1.069
p-value (Wu-Hausman F-test)	0.387	0.524	0.380	0.395	0.301
First Stage F-test	16.583	14.746	16.955	16.988	47.467
p-value (First Stage F-test)	0.000	0.000	0.000	0.000	0.000
Observations	2178	2178	2178	2178	2174

Panel A: Loan Maturity (months)

×	(1)	(2)	(3)	(4)	(5)
	IV Probit	IV Probit	IV Probit	IV Probit	Post-LASSO
Diffusion		0.11/***			0.172***
Diffusion		-0.114			(0.032)
Sourcing		-0 113***			(0.032)
Sourcing		(0.040)			
Degree		(0.010)	-0.049*		
Degree			(0.027)		
Eigenvector			(0.027)	-0.052*	
Ligenvector				(0.022)	
Ln(Lead-PE Relationship)	0.015	0.014	0.022	0.021	0.014
F)	(0.015)	(0.016)	(0.016)	(0.016)	(0.013)
Ln(Lead-Target Relationship)	0.056**	0.035	0.061***	0.061***	0.035*
F/	(0.023)	(0.025)	(0.024)	(0.023)	(0.021)
Avg Number of Members	0.142***	0.155***	0.133***	0.138***	0.167***
6	(0.038)	(0.039)	(0.038)	(0.038)	(0.029)
Non US Lead	-0.328***	-0.283***	-0.304***	-0.309***	-0.280***
	(0.078)	(0.085)	(0.079)	(0.079)	(0.070)
Ln (Deal Val)	0.289***	0.295***	0.282***	0.284***	0.317***
	(0.046)	(0.047)	(0.046)	(0.046)	(0.039)
Captive PE	0.193	0.153	0.203	0.206	()
- II I	(0.192)	(0.205)	(0.193)	(0.192)	
Ln (PE Age)	0.095*	0.093	0.090	0.088	
	(0.057)	(0.060)	(0.057)	(0.057)	
Lead LBO Mkt Share	-0.010***	-0.009***	-0.011***	-0.011***	
	(0.003)	(0.003)	(0.003)	(0.003)	
Ln (Target Assets)	-0.239*	-0.260*	-0.246*	-0.248*	-0.346***
	(0.129)	(0.133)	(0.130)	(0.129)	(0.102)
Ln (Target EBITDA)	-1.346	-1.077	-1.396	-1.365	
	(1.250)	(1.282)	(1.253)	(1.247)	
Target LT Debt to Assets	-0.927*	-1.103**	-0.905*	-0.877	
0	(0.540)	(0.555)	(0.542)	(0.540)	
No Compustat Data	-0.331**	-0.226	-0.326**	-0.324**	
*	(0.160)	(0.167)	(0.161)	(0.160)	
Syndicate Herfindahl	-0.366**	-0.175	-0.358**	-0.367**	-0.413***
•	(0.166)	(0.179)	(0.167)	(0.166)	(0.138)
Ln (Loan Maturity)	2.083***	2.049***	2.093***	2.044***	1.736***
	(0.492)	(0.558)	(0.492)	(0.490)	(0.485)
Constant	-6.850***	-6.835**	-6.769***	-6.572***	-6.643***
	(2.537)	(2.830)	(2.539)	(2.533)	(2.524)
Target Industry FE	Yes	Yes	Yes	Yes	Yes
Time FE	Yes	Yes	Yes	Yes	Yes
χ²-test	364.475	363.360	368.747	369.641	447.010
p-value (χ^2 -test)	0.000	0.000	0.000	0.000	0.000
Wald test of weak instruments	24.928	18.653	25.325	24.010	15.959
p-value Wald test	0.000	0.000	0.000	0.000	0.000
Observations	2174	1998	2174	2174	2174

I ANELD. I LUDADINU VI CUNALELAI VILLIDVI IVANS

$\begin{array}{c c c c c c c c c c c c c c c c c c c $		(1)	(2)	(3)	(4)	(5)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $		IV	IV	IV	IV	Post-LASSO
(0.033)(0.020)Sourcing -0.093^{***} -0.055^{**} (0.020) (0.023) (0.023) Degree -0.118^* (0.068) Eigenvector -0.011^{**} (0.080) Ln(Bank-PE Relationship) -0.001 -0.052^{**} -0.066^{**} (0.021) (0.021) (0.032) (0.035) Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.115^{***} (0.026) (0.024) (0.033) (0.034) Avg Number of Members -0.194^{***} -0.151^{***} -0.131^{**} (0.046) (0.041) (0.044) (0.052) Non US Lead 0.043 0.006 0.005 -0.066 (0.111) (0.107) (0.121) (0.143) Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 (0.056) (0.057) (0.061) (0.074) (0.049) Captive PE -0.132 0.040 0.039 0.053	Diffusion		-0.093***			-0.087***
Sourcing -0.093^{***} -0.055^{**} Degree -0.118^* (0.023) Eigenvector -0.191^{**} (0.068) Ln(Bank-PE Relationship) -0.001 -0.052^{**} -0.066^{**} -0.088^{**} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.115^{***} -0.132^{***} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.155^{***} -0.132^{***} (0.026) (0.024) (0.033) (0.034) Avg Number of Members -0.194^{***} -0.151^{***} -0.131^{**} -0.178^{***} Non US Lead 0.043 0.006 0.005 -0.066 (0.111) (0.107) (0.121) (0.143) Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 -0.103^{**} Captive PE -0.132 0.040 0.039 0.053 (0.049) (0.126) (0.128) (0.136) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049) (0.049)			(0.033)			(0.020)
(0.032) (0.023) Degree -0.118^* (0.068) Eigenvector -0.191^{**} (0.080) Ln(Bank-PE Relationship) -0.001 -0.052^{**} -0.066^{**} -0.088^{**} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.115^{***} -0.132^{***} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.155^{***} -0.132^{***} (0.026) (0.024) (0.033) (0.034) Avg Number of Members -0.194^{***} -0.151^{***} -0.131^{**} -0.178^{***} (0.046) (0.041) (0.044) (0.052) (0.032) Non US Lead 0.043 0.006 0.005 -0.066 (D.111) (0.107) (0.121) (0.143) Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 -0.103^{**} Captive PE -0.132 0.040 0.039 0.053 (0.049) (0.126) (0.128) (0.136)	Sourcing		-0.093***			-0.055**
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	C		(0.032)			(0.023)
(0.068) Eigenvector $-0.191**$ (0.080) Ln(Bank-PE Relationship) -0.001 -0.052^{**} -0.066^{**} -0.088^{**} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.115^{***} -0.132^{***} Ln(Bank-Target Relationship) -0.056^{**} -0.099^{***} -0.115^{***} -0.132^{***} (0.026) (0.024) (0.033) (0.034) Avg Number of Members -0.194^{***} -0.151^{***} -0.131^{**} -0.178^{***} (0.046) (0.041) (0.044) (0.052) (0.032) Non US Lead 0.043 0.006 0.005 -0.066 Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 -0.103^{**} Captive PE -0.132 0.040 0.039 0.053 (0.140) (0.126) (0.128) (0.0136)	Degree		. ,	-0.118*		
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	-			(0.068)		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Eigenvector				-0.191**	
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	-				(0.080)	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Ln(Bank-PE Relationship)	-0.001	-0.052**	-0.066**	-0.088**	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(0.021)	(0.021)	(0.032)	(0.035)	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Ln(Bank-Target Relationship)	-0.056**	-0.099***	-0.115***	-0.132***	
Avg Number of Members -0.194^{***} -0.151^{***} -0.161^{***} -0.131^{**} -0.178^{***} Non US Lead 0.046) (0.041) (0.044) (0.052) (0.032) Non US Lead 0.043 0.006 0.005 -0.066 (0.111) (0.107) (0.121) (0.143) Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 -0.103^{**} Captive PE -0.132 0.040 0.039 0.053 (0.140) (0.126) (0.128) (0.136)		(0.026)	(0.024)	(0.033)	(0.034)	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Avg Number of Members	-0.194***	-0.151***	-0.161***	-0.131**	-0.178***
Non US Lead 0.043 0.006 0.005 -0.066 (0.111) (0.107) (0.121) (0.143) Ln (Deal Val) -0.196*** -0.089 -0.090 -0.048 -0.103** (0.056) (0.057) (0.061) (0.074) (0.049) Captive PE -0.132 0.040 0.039 0.053		(0.046)	(0.041)	(0.044)	(0.052)	(0.032)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Non US Lead	0.043	0.006	0.005	-0.066	
Ln (Deal Val) -0.196^{***} -0.089 -0.090 -0.048 -0.103^{**} (0.056)(0.057)(0.061)(0.074)(0.049)Captive PE -0.132 0.0400.0390.053(0.140)(0.126)(0.128)(0.136)		(0.111)	(0.107)	(0.121)	(0.143)	
Captive PE (0.056) (0.057) (0.061) (0.074) (0.049) (0.140) (0.126) (0.128) (0.136)	Ln (Deal Val)	-0.196***	-0.089	-0.090	-0.048	-0.103**
Captive PE -0.132 0.040 0.039 0.053 (0.140) (0.126) (0.128) (0.136)		(0.056)	(0.057)	(0.061)	(0.074)	(0.049)
(0.140) (0.126) (0.128) (0.136)	Captive PE	-0.132	0.040	0.039	0.053	
(0.120) (0.120) (0.150)		(0.140)	(0.126)	(0.128)	(0.136)	
Ln (PE Age) 0.098* 0.047 0.045 0.044	Ln (PE Age)	0.098*	0.047	0.045	0.044	
$(0.060) \qquad (0.053) \qquad (0.053) \qquad (0.053)$		(0.060)	(0.053)	(0.053)	(0.053)	
Lead LBO Mkt Share 0.000 0.000 0.000 0.000	Lead LBO Mkt Share	0.000	0.000	0.000	0.000	
(0.000) (0.000) (0.000) (0.000) (0.000)		(0.000)	(0.000)	(0.000)	(0.000)	
Ln (Target Assets) -0.332^* -0.261 -0.293 -0.417^* -0.042	Ln (Target Assets)	-0.332*	-0.261	-0.293	-0.417*	-0.042
$(0.183) \qquad (0.198) \qquad (0.194) \qquad (0.241) \qquad (0.136)$		(0.183)	(0.198)	(0.194)	(0.241)	(0.136)
Ln (Target EBITDA) -1.189 -2.261 -2.101 -1.269	Ln (Target EBITDA)	-1.189	-2.261	-2.101	-1.269	
$(1.965) \qquad (1.687) \qquad (1.575) \qquad (1.773)$		(1.965)	(1.687)	(1.575)	(1.773)	0.045
Target LT Debt to Assets 1.102^* -0.073 -0.045 0.155 0.345	Target LT Debt to Assets	1.102*	-0.073	-0.045	0.155	0.345
(0.623) (0.381) (0.391) (0.420) (0.557)		(0.623)	(0.381)	(0.391)	(0.420)	(0.557)
No Compustat Data 0.424^{**} 0.485^{**} 0.521^{***} 0.584^{***}	No Compustat Data	0.424**	0.485**	0.521***	0.584***	
$(0.198) \qquad (0.199) \qquad (0.189) \qquad (0.203) $		(0.198)	(0.199)	(0.189)	(0.203)	0.700
Ln (Loan Maturity) $0.191 - 0.036 - 0.108 - 1.762 - 0.798$	Ln (Loan Maturity)	0.191	-0.036	-0.108	-1.762	-0.798
(1.722) (2.188) (2.097) (2.847) (2.036)		(1./22)	(2.188)	(2.097)	(2.847)	(2.036)
Loan Collateral -2.253^{++} -1.406 -1.439 -0.894 $-1.17/$	Loan Collateral	-2.253**	-1.406	-1.459	-0.894	-1.1//
(0.369) (0.948) (0.905) (1.242) (0.790)	A	(0.889)	(0.948)	(0.905)	(1.242)	(0.790)
Avg on spread $(0.482^{-100} - 0.59)^{-100} = (0.50^{-100} - 0.50^{-100$	Avg om spread	(0.482^{***})	(0.101)	(0.005)	(0.120)	0.5/1***
$\begin{pmatrix} (0.102) & (0.101) & (0.095) & (0.129) & (0.094) \\ 0.555 & 0.624 & 0.226 & 16.770 & 7.167 \\ \end{pmatrix}$	Constant	(0.102)	(0.101)	(0.093)	(0.129)	(0.094)
Constant 6.305 6.054 9.350 10.779 7.107	Constant	6.303 (7.426)	8.034 (0.427)	9.550	(12,528)	(8.060)
(7.420) (9.457) (9.525) (12.526) (8.500)	Torrat Industry FE	(7.420) Vas	(9.457) Vas	(9.525) Vac	(12.526)	(8.900) Vac
Time EE Van Van Van Van Van	Time EE	T es Vos	Tes Vac	Vas	Vas	Vas
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Adjusted \mathbf{P}^2	0.224	0.314	0.311	0.206	0.261
Adjusted 0.224 0.514 0.511 0.200 0.201 Durbin v2-toet 9.413 5.678 4.013 5.741 0.537	Aujusicu R Durbin χ^2 -tost	8 4 4 3	5.678	4 013	5 741	0.201
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$p_{\rm value}$ (Durbin χ^2 -tost)	0.015	0.058	4.915	0.057	0.765
p -value (Dation $\chi^{-1}(S_{1})$ 0.015 0.056 0.060 0.007 0.005 Wu-Hansman E-test 4.081 2.736 2.358 2.768 0.261	μ -value (Durbin χ test) Wu-Hausman E-test	4 081	2 736	2 368	2 768	0.765
rue-rausman rusa +.001 2.750 2.500 2.700 0.201	n-value (Wu-Hausman F-test)	0.017	2.750	0.094	0.063	0.201
First Store Etect 14 396 11 863 13 705 11 050 13 572	First Stage F-test	14 396	11 863	13 705	11 950	13 572
nyalue (First Stage F-test) 0,000 0,000 0,000 0,000 0,000	n-value (First Stage F-test)	0.000	0.000	0.000	0.000	0.000
Observations 2115 2115 2115 2115	Observations	2115	2115	2115	2115	2115

Panel C: LBO loan spread (excess of LIBOR)

Lending when Relationships are Scarce: The Role of Information Spread via Bank Networks

Yan Alperovych[†], Anantha Divakaruni[‡], and Sophie Manigart^{\perp}

INTERNET APPENDIX

Appendix A. Pairwise correlations between the lead and member bank centralities	1
Appendix B: LBO characteristics with/without bank-borrower ties	2
Appendix C. Market share of banks in the US LBO loan market	7
Appendix D. Description of the propensity matching procedure	8
Appendix E. Economic effect: Lead bank centralities and LBO loan terms	12

[†] Emlyon Business School; alperovych@em-lyon.com

[‡] University of Bergen; Email: anantha.divakaruni@uib.no

 $^{^{\}perp}$ Vlerick Business School and Ghent University; sophie.manigart@vlerick.com

Appendix A. Pairwise correlations between the lead and member bank centralities

Table presents pairwise correlations between network centralities of lead and syndicate member banks that were active in the US leveraged buyout (LBO) loan market during the period 1991-2012. A US LBO loan refers to syndicated loans sponsored by US-based PE firms to acquire US-based targets. See Table 1 for variables description.

	DiffusionLead	SourcingLead	DiffusionMember	Sourcing _{Member}
DiffusionLead	1			
SourcingLead	-0.056**	1		
Diffusion _{Member}	0.964***	-0.089***	1	
Sourcing _{Member}	-0.105***	0.919***	-0.099***	1

Appendix B: LBO characteristics with/without bank-borrower ties

Figures present a visual comparison of LBO lending terms and syndicate characteristics between deals. Red dots (and quadratic fit plots) represent deals where the lead bank(s) has prior relationships with LBO borrowers (i.e. PE firm or LBO target depending on the specification). Blue dots (and quadratic fit plots) represent deals where the lead bank(s) did not have any prior relationship with the LBO borrowers. LBO deal size on the X-axis is restricted to \$1.5 billion for brevity. Bank–borrower relationships are measured in five-year rolling windows preceding a given deal year.

See Table 1 for variables description.

A) LBO deal size versus loan maturity (in months)





B) LBO deal size versus collateral requirements



C) LBO deal size versus loan interest rate spread (excess of LIBOR)

No Bank-PE Relation Thip
 Bank-PE Relation Thip

No Bank-Target RelationThip
 Bank-Target RelationThip

D) LBO deal size versus loan syndicate size



5

E) LBO deal size versus lead bank share (%)



6

Appendix C. Market share of banks in the US LBO loan market

Figure shows the market share of individual banks active in the US LBO loan market by year. A bank is considered active in year t if it was involved in at least one LBO in the preceding five years, i.e. between t-5 and t. Bank market share is the percentage of all LBO loans (in mil. USD) during the previous five years in which the bank participated either as lead or member. Blue line represents banks that are domiciled or headquartered in the US. Red line represents banks domiciled or headquartered outside the US.



Appendix D. Description of the propensity matching procedure

To answer whether a bank's decision to join an LBO syndicate is influenced by the information it receives (diffuses) from (to) the network, and by its topological position within the network, we ideally require data on banks that joined LBO syndicates and also those that were invited but did not join eventually. However, our primary data source, LPC Dealscan, reports only completed deals and does not provide data on banks that opted out of joining, or were not invited to join, an LBO syndicate. To overcome this limitation and develop requisite counterfactuals for analysis, we proceed in two steps.

In the first step, for each deal, we construct a block comprising incumbent nonlead syndicate members and all other banks that were active in the LBO loan market during the five preceding years. Together, these blocks represent a deal-level sample comprising both eventual (*case*) and potential (*control*) syndicate members. This case-control sample contains 751,977 observations. However, using this sample for analysis would require an implicit assumption that the mean lead bank invites up to 324 prospective banks to join its LBO loan syndicate (which ultimately comprise only 2-3 members on average). This assumption is unrealistic from a cost and time perspective. A more realistic approach to syndication is that the lead bank first approaches banks with whom it is least likely to face adverse selection problems. This is most likely the case with prospective banks that are closest to the lead bank in terms of individual characteristics and lending activity, and possibly collaborated with it in previous deals.

Accordingly, in the second step, we construct a more realistic case-control sample of LBO syndicate joining using a propensity score (PS) matching approach (Dehejia and Wahba, 2002; Rosenbaum and Rubin, 1983). In particular, within each deal-level block from the case-control sample, we identify up to ten *control* banks that are closest to incumbent nonlead *case* members in observable characteristics, and consequently closest in terms of their syndicate joining probability. The PS matching is done on three observable bank characteristics: (i) prior experience in arranging LBO loan syndicates (*Bank Lead Experience*), (ii) reputation (*Bank LBO Market Share*), and (iii) domicile (*Non US Bank*). *Bank Lead Experience* is defined as the percentage of participated LBO loans in the last five years in which the bank was the lead arranger. Bank LBO Market Share is the percentage of all LBO loans (in mil. USD) during the previous five years in which the bank participated either as lead or member. Finally, US Bank is a dummy equal to one if the bank is domiciled or headquartered outside the US.

The propensity scores are computed using the conditional logit (CL) model with deal-level clustering as specified below:

$\Lambda(\text{Syndicate member}_{ij}=1) = \beta_1 \text{Bank Lead Experience}_{ij} + \beta_2 \text{Bank LBO Market Share}_{ij} + \beta_3 \text{Non US Bank}_i + \epsilon_{ij},$

where *i* refers to an LBO, and *j* refers to a bank that can potentially join the syndicate. The dependent variable Λ (Syndicate member_{*ij*}=1) is the log-odds of joining the LBO syndicate as nonlead member. Upon fitting the model, we predict each bank's probability (*P*) to join the syndicate and compute the difference in *P* (denoted Δ) between each case and controls bank in a deal. Next we rank the control banks in ascending order by the absolute value of $|\Delta|$ and retain the top 10 banks. This yields *N*+10 banks per deal, where *N* is the number of actual members (cases) in a syndicate.³⁵ All our matching is done with replacement to reduce bias in the estimates (Abadie and Imbens, 2002).

Figure D.1 presents the distribution of the difference in $P(\Delta)$ in the resulting case-control sample. Panel A shows the distribution of Δ between case banks and their first best matches (control banks that are the closest to cases in terms of syndicate joining probability). Panels B–D displays the distribution of Δ of the first three, five, and ten nearest matches, respectively. The red vertical lines denote the 5th and 95th percentiles of the distribution of Δ . The distributions are centered around zero with over 90% matches falling within the range indicated by the red lines.

Figures D.2–D.4 present distributions of the differences in observable covariates between the case and control banks. Once again, the distributions are centered around zero, except for some visible skew in the case of *Non US Bank* dummy. We compute the proportion of matches that fall within the in 5th and 95th percentile range for each of the continuous covariates (indicated by red dotted lines). For *Bank Lead Experience* the average proportion (average across panels A-D) is 90.03% and for *Bank Market Share* this proportion is 90.09%.

Overall, these results suggest that the PS matching procedure is able to identify valid counterfactual banks for our analysis.

³⁵ Note that using the more canonical approach of matching ten closest candidates to each *case* bank would yield $N + N \times 10$ case-control banks per deal. However, this would imply that the lead bank invites up to 30 prospective candidates to form the mean LBO syndicate comprising 2-3 members, which is implausible from a cost and time perspective as argued above.

Figure D.1: Differences in *P* between LBO syndicate members (*cases*) and matched banks (*controls*)



Figure D.2: Differences in *Bank Lead Experience* between LBO syndicate members (*cases*) and matched banks (*controls*)



Figure D.3: Differences in *Bank Market Share* between LBO syndicate members (*cases*) and matched banks (*controls*)



Figure D.4: Differences in *Non US bank* dummy between LBO syndicate members (*cases*) and matched banks (*controls*)



Appendix E. Economic effects: Lead bank centralities and LBO loan terms

Table reports the expected change in PE equity returns for a one-standard-deviation increase in lead bank network centralities and other control variables and the corresponding change in LBO lending terms as reported in Table 9.

To estimate these effects, we make several assumptions about the deal structure and maturity of LBO loans. First, following Guo et al. (2011), we assume that the median equity contribution in LBO deals is around 30% of the transaction price. Second, based on data from ThomsonOne, we assume that the median debt contribution in an LBO financed through syndicated loans is approximately 46% of the transaction price. This implies that for an average syndicated loan package of size \$303 million, the corresponding equity contribution by PE firms is around \$198 million. Third, assuming that PE firms exit their investments (LBO targets) at the end of the loan term and do not seek any refinancing, we use the average loan maturity of 5.7 years as a proxy for the average holding period of LBO investments.

Based on these assumptions, we estimate the economic significance of our explanatory variables through the following steps. First, we multiply each coefficient reported in Table 9 by the standard deviation of the corresponding explanatory variable to estimate the change in annual repayment against the average LBO loan value of \$303 million. Second, we discount these changes in annual repayment over the average holding period of 5.7 years at a base interest rate of 5.09% (12-month LIBOR of 1.25% + average LBO loan spread of 3.84%) and aggregate them to produce the present value of cash savings. Third, we compare these aggregate discounted cash savings against the average equity contribution of \$198 million to determine the change in return on investment (internal rate of return) for PE firms. Since a common set of controls is used across all models, we report the average change in PE equity returns for a one-standard-deviation increase in these variables. A detailed description of all variables is available in Table 1.

Panel A: Loan Maturity

								Excess
						Change in		PE
					Annual Loan	Annual		Equity
					Repayment	Repayment	PV	Return
Variables	Coefficient	Mean	SD	1 SD up	(\$m)	(\$m)	(Annuity)	(%)
Bank network centralities					62.76 (Base)			
Diffusion (2)	0.017***	0.8	1.4	0.02	62.74	0.02	0.09	0.05
Sourcing (2)	0.005	0.4	1.2	0.01	62.76	0.00	0.02	0.01
Degree (3)	-0.001	0.9	1.3	0.00	62.76	0.00	-0.01	0.00
Eigenvector (4)	0.003	0.9	1.2	0.00	62.76	0.00	0.01	0.01
Average coef. (models 1-4)								
Ln(Bank-PE Relationship)	0.002	4.68	5.86	0.01	62.75	0.01	0.05	0.03
Ln(Bank-Target Relationship)	-0.001	2.66	4.36	-0.01	62.77	0.00	-0.02	-0.01
Bank LBO Mkt Share	0.000	1.01	0.00	0.00	62.76	0.00	0.00	0.00
Non US Bank	0.036	0.70	0.50	0.02	62.75	0.01	0.07	0.04
Ln (PE Age)	-0.010	2.93	2.74	-0.03	62.78	-0.02	-0.10	-0.05
Ln (Deal Val)	-0.016	5.72	5.98	-0.09	62.84	-0.07	-0.36	-0.18
Loan Collateral	0.091	0.50	0.50	0.05	62.73	0.04	0.18	0.09

Panel B: Loan Collateral (D)

				One SD up (change in probability
Variables	Coefficient	Mean	SD	of pledging collateral)
Bank LBO centralities				
Diffusion (2)	-0.114***	0.8	1.4	-0.16
Sourcing (2)	-0.113***	0.4	1.2	-0.13
Degree (3)	-0.049*	0.9	1.3	0.07
Eigenvector (4)	-0.052*	0.9	1.2	-0.06
Average coef (models 1-4)				
Ln(Bank-PE Relationship)	0.018	4.68	5.86	0.11
Ln(Bank-Target Relationship)	0.053	2.66	4.36	0.23
Bank LBO Mkt Share	-0.010	1.01	1.36	-0.01
Non US Bank	-0.306	0.70	0.40	-0.12
Ln (PE Age)	0.092	2.93	2.74	0.25
Ln (Deal Val)	0.288	5.72	5.98	1.72
Ln (Loan Maturity)	2.067	4.20	2.69	5.57

Б

Panel C: Loan Spread

								Excess
					Annual	Change in		PE
					Loan	Annual		Equity
					Repayment	Repayment	PV	Return
Variables	Coefficient	Mean	SD	1 SD up	(\$m)	(\$m)	(Annuity)	(%)
Bank network centralities					62.76 (Base)			
Diffusion (2)	-0.093***	0.8	1.4	-0.13	62.52	0.25	1.19	0.60
Sourcing (2)	-0.093***	0.4	1.2	-0.109	62.55	0.21	1.02	0.51
Degree (3)	-0.118*	0.9	1.3	-0.16	62.46	0.30	1.47	0.74
Eigenvector (4)	-0.191**	0.9	1.2	-0.23	62.31	0.45	2.19	1.11
Average coef. (models 1-4)								
Ln(Bank-PE Relationship)	-0.052	4.68	5.86	-0.30	62.18	0.58	2.84	1.43
Ln(Bank-Target Relationship)	-0.101	2.66	4.36	-0.44	61.92	0.84	4.11	2.08
Bank LBO Mkt Share	0.000	1.01	1.36	0.00	62.76	0.00	0.00	0.00
Non US Bank	-0.003	0.70	2.69	-0.01	62.75	0.02	0.08	0.04
Ln (PE Age)	0.059	2.93	2.74	0.16	63.07	-0.31	-1.48	-0.75
Ln (Deal Val)	-0.106	5.72	0.50	-0.05	62.66	0.10	0.49	0.25
Ln (Loan Maturity)	-0.429	4.20	2.69	-1.16	60.56	2.20	11.03	5.57
Loan Collateral	-1.503	0.5	0.5	-0.75	61.33	1.44	7.11	3.59